

Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17911

Gender Norms and Female Labor Supply: Evidence from Export Shocks in Vietnam

Quynh Huynh Hyejin Ku

MAY 2025



Initiated by Deutsche Post Foundation

DISCUSSION PAPER SERIES

IZA DP No. 17911

Gender Norms and Female Labor Supply: Evidence from Export Shocks in Vietnam

Quynh Huynh University College London and CReAM

Hyejin Ku University College London, CEPR, IZA and RFBerlin

MAY 2025

Any opinions expressed in this paper are those of the author(s) and not those of IZA. Research published in this series may include views on policy, but IZA takes no institutional policy positions. The IZA research network is committed to the IZA Guiding Principles of Research Integrity.

The IZA Institute of Labor Economics is an independent economic research institute that conducts research in labor economics and offers evidence-based policy advice on labor market issues. Supported by the Deutsche Post Foundation, IZA runs the world's largest network of economists, whose research aims to provide answers to the global labor market challenges of our time. Our key objective is to build bridges between academic research, policymakers and society.

IZA Discussion Papers often represent preliminary work and are circulated to encourage discussion. Citation of such a paper should account for its provisional character. A revised version may be available directly from the author.

ISSN: 2365-9793

IZA – Institute of Labor Economics

Schaumburg-Lippe-Straße 5–9 53113 Bonn, Germany	Phone: +49-228-3894-0 Email: publications@iza.org	www.iza.org
		5

ABSTRACT

Gender Norms and Female Labor Supply: Evidence from Export Shocks in Vietnam^{*}

We examine the relationship between economic development and female labor force participation, with a focus on the impact of gender norms. Analyzing quasi-random variation in provincial exports in reunified Vietnam from 2002 to 2018, we find that a positive economic shock led to a significant decline in women's labor market engagement, particularly among married women from wealthier households and those with husbands in more skilled occupations. This trend is more pronounced in the South (formerly capitalist) than in the North (always socialist), and among native Southerners compared to Northerners relocated to the South after the war. Our findings highlight the importance of gender role attitudes in shaping women's responses to rising incomes.

JEL Classification:	J16, J22, O12
Keywords:	female labor force participation, social norms, gender role
	attitudes, income and substitution effects, trade liberalization

Corresponding author:

Hyejin Ku Department of Economics University College London 30 Gordon Street London, WC1H 0AX United Kingdom E-mail: h.ku@ucl.ac.uk

^{*} We thank the Editor, anonymous referees, Jerome Adda, Kirill Borusyak, Eric Edmonds, Julian Costas- Fernandez, Swati Dhingra, Christian Dustmann, Bernd Fitzenberger, Guido Friebel, Albrecht Glitz, Ararat Gocmen, Attila Lindner, Matt Nibloe, Suphanit Piyapromdee, Anna Raute, Uta Schoenberg, Michel Serafinelli, Takahiro Toriyabe, Joonas Tuhkuri, Marcos Vera-Hernandez, and seminar and conference participants at AASLE, CReAM-UCL, EALE, Leibniz University Hannover, RFBerlin, and Tsukuba University for helpful comments and discussions. We gratefully acknowledge financial support from the European Research Council under the European Union's Horizon 2020 research and innovation program (grant agreement no. 818992) and the British Academy/Leverhulme Small Research Grant SRG2324/241828.

1. Introduction

The relationship between economic development and female labor force participation (FLFP) is an important topic that has received considerable attention (see, e.g., Goldin, 1995; Mammen and Paxson, 2000; Blau and Kahn, 2013; Olivetti, 2013; Gaddis and Klasen, 2014; Ngai and Petrongolo, 2017; Afridi et al., 2018; Klasen, 2019; Dinkelman and Ngai, 2022; Ngai et al., 2024). While many studies focus on testing the U-shaped relationship between FLFP and income (Boserup, 1970; Goldin, 1995), they also reveal notable variations in FLFP among countries with similar income levels, highlighting the influence of additional factors on women's economic engagement.

In this paper, we explore how economic growth affects women's labor market participation, emphasizing the influence of social norms. We examine the context of reunified Vietnam, particularly during the trade liberalization of the 2000s and its subsequent rapid growth. Following the US-Vietnam Bilateral Trade Agreement in 2002 and Vietnam's entry to the World Trade Organization in 2007, the country has been experiencing rapid export expansion (see Figure A1 and Table A1) and large inflows of foreign direct investment (FDI). Our investigation centers on how exogenous increases in economic opportunities affect women's labor market engagement and how these effects are moderated by prevailing gender role attitudes (e.g., the male breadwinner model vs. gender-egalitarian perspectives).

Vietnam serves as a compelling case study for analyzing shifts in female labor supply in the context of economic development. With a female-to-male labor force participation ratio exceeding 90%, Vietnam has one of the smallest gender gaps in labor force participation (LFP) worldwide, surpassing even advanced economies such as Sweden, the US, the UK, and Germany (Figure 1). However, since the 2000s, the gender gap in LFP has begun to widen, coinciding with rapid export expansion and rising incomes. Given the ongoing policy

1

discussions aimed at retaining female workers in the labor market, it is important to understand the conditions under which women's LFP may decline as incomes rise.

[Figure 1]

Our analysis draws on biennial microdata from the Vietnam Household Living Standards Surveys (VHLSS) for the period 2002–2018, combined with trade data from the United Nations Comtrade Database. To proxy for the exogenous progress of the local economy, we use predicted exports by province and year based on a Bartik instrument, allocating nationwide industry-specific exports to different provinces according to their initial (i.e., pre-2002) industry composition, similar to the approach used by Topalova (2010), Autor et al. (2013), and Kovak (2013). We then relate our outcomes of interest, such as individuals' working status, to the province-year level predicted exports within a difference-in-differences (DID) framework, conditioning on province FE and year FE. Although data on actual exports at the province-by-year level are not available, we confirm that predicted exports by province are significantly and positively correlated with gross regional domestic product (GRDP) per capita, available for 2010-2018.

Our main findings show that an increase in predicted provincial exports—referred to as "export shock" or "export exposure" for simplicity—leads to a significant decrease in women's employment, while men's employment remains largely unaffected. Specifically, provinces with one standard deviation (SD) higher export shock experience a decline in women's working status by 2.7 percentage points (pp). This shift translates to a widening of the gender gap in LFP by 3 pp (or 56% of the mean gender gap).

One might suspect that the decline in women's employment is due to demand-driven displacement from trade liberalization, but our analysis does not support this. We find that export shocks promote structural changes in labor allocation from agriculture to the manufacturing and services sectors. Additionally, these shocks shift workers into the formal sector, aligning with findings from McCaig and Pavcnik (2015, 2018).¹ Importantly, female workers benefit more than male workers from expanding economic opportunities, with a swifter transition out of agriculture and into manufacturing and services. Specifically, a one SD increase in export exposure disproportionately raises women's employment (relative to men) in manufacturing by 1.4 pp (9.3% of the female average) and in services by 1.8 pp (5.5% of the female average). This trend corresponds with the rise of foreign direct investment in low-skilled, labor-intensive sectors in Vietnam. Overall, the evidence suggests that the decline in women's participation is not due to adverse labor demand shocks disproportionately affecting them.

Why would Vietnamese women reduce their labor supply amid expanding economic opportunities? We hypothesize that this relates to the balance between income and substitution effects, as well as household labor specialization. Our analysis reveals that the decline in female labor supply is driven entirely by married women, while single women display patterns similar to men. Among married women, the negative impact of export shocks is most pronounced among those from wealthier households and those with husbands in more skilled occupations, where the income effect (encouraging women to focus on home production) is more likely to outweigh the substitution effect (encouraging market work due to high opportunity costs) resulting from the export shock.

Based on these findings, we further hypothesize that whether income or substitution effects dominate for the same increase in income may depend on the prevailing gender role attitudes in society. Specifically, we anticipate that, when faced with rising incomes, women's

¹ McCaig and Pavcnik (2015, 2018) demonstrate that within just two years of the US-Vietnam Bilateral Trade Agreement (UVBTA) in December 2001, reductions in US tariffs on Vietnamese exports facilitated the reallocation of manufacturing workers to the formal sector, resulting in larger, more productive firms and higher wages. Our findings align with these effects, revealing similar patterns of export-induced formalization of employment based on nearly two decades of data (2002–2018). Recent research by McCaig et al. (2023) supports these conclusions. Additionally, we document a trade-induced structural shift from agriculture to manufacturing and services.

labor market attachment may be stronger in societies with gender-egalitarian norms than in those with traditional male breadwinner ideals.

Vietnam serves as a valuable context for this exploration. Before reunification in 1976, the northern and southern regions were governed by vastly different ideologies. During the Vietnam War (1954-1975), North Vietnam promoted gender equality and women's economic participation, while South Vietnam did not (Werner, 1981). Nearly 40 years after reunification, women's attachment to the labor market still varies significantly between the North and South: the LFP rate for women in the North is as high as 90%, comparable to that of men, while in the South it is around 80% (see Figure 2). Data from the World Values Survey further support this divergence in gender role attitudes, with Northern Vietnamese of both genders exhibiting more gender egalitarian perspectives.

Our analysis reveals significant regional differences in the impact of export shocks on FLFP. While women in both regions reduce their labor supply, the decline is more pronounced in formerly capitalist South Vietnam compared to socialist North Vietnam. Unlike Northern Vietnamese, Southerners experienced socialism more recently and to a lesser extent.² Consequently, Southern Vietnamese may be more susceptible to the resurgence of the male breadwinner model, influenced by trade liberalization and the growing presence of capitalism.³ In the North, the negative impact on female labor supply is primarily observed among women married to husbands in the top earnings quartile. This gradient is absent in the South, where women reduce their labor supply across all quartiles of household wealth and husbands' earnings. This suggests that women's labor market attachment is weaker in the South,

² Since the Doi Moi reforms in 1986, Vietnam has transitioned from a centrally planned economy to a market economy, marking a departure from the post-reunification "lost decade." Despite being governed by a socialist regime since reunification in 1976, the influence of communist ideologies has significantly weakened (see, e.g., Werner and Bélanger, 2002; Le, 2012; Tarp, 2018).

³ As Vietnam liberalizes its trade policies and shifts towards a market economy, its economic structure increasingly resembles the capitalist model that was predominant in former South Vietnam.

prompting them to withdraw from the workforce as soon as it becomes financially feasible. In contrast, Northern women demonstrate stronger labor market attachment, with the income effect becoming dominant only at a higher household income threshold.

Our finding that the trade-induced FLFP reduction is stronger in South Vietnam than in North Vietnam remains stable when excluding provinces in the Mekong River Delta or those historically part of the Khmer Empire, whose institutions are distinct from the rest of Vietnam (see Dell et al., 2018). We also use provincial exports predicted separately based on maleintensive and female-intensive industries to account for potential regional differences in economic structure. The results remain consistent with our baseline findings. This ensures that the North-South differences in FLFP effects are not driven by variations in the composition of exporting industries across regions, particularly regarding the gender-friendliness of the jobs created.

The North-South differences in behavior may reflect not only the influence of past political regimes and wartime experiences but also differences in other dimensions such as geography or institutional factors. To keep such environmental conditions constant, we leverage the forced migration policy implemented in the decade following reunification, known as the New Economic Zones (NEZ) program. Enacted by the Vietnamese Communist Party for both economic and political reasons (Desbarats, 1987; Dang et al., 1997), the policy relocated millions of Northerners to previously uninhabited mountainous or rural areas that formerly belonged to South Vietnam. Because these migrants were born and raised in the North, their exposure to the socialist regime is longer than that of native Southerners currently residing in the same provinces.⁴

⁴ See, e.g., Antecol (2000), Fisman and Miguel (2007), Giuliano (2007), Fernández and Fogli (2009), and Blau et al. (2011), for studies using variation in immigrants' country of origin to identify the role of culture and social norms.

Our findings indicate that, in response to provincial export shocks, the labor supply responses of the North-born population living in the South closely resemble those of individuals in the North, highlighting the lasting influence of wartime values. These results hold even in our most restrictive specifications, where we directly compare North-born and South-born migrants, confirming that the effects observed among individuals from the North align with those of their Northern counterparts and are not merely a function of migration status.

By examining women's labor supply responses to Vietnam's rapid export expansion from 2002 to 2018, we provide a new perspective on understanding the relationship between economic development and women's labor market engagement (see e.g., Boserup, 1970; Goldin, 1995; Mammen and Paxson, 2000; Blau and Kahn, 2013; Olivetti, 2013; Gaddis and Klasen, 2014; Ngai and Petrongolo, 2017; Klasen, 2019; Dinkelman and Ngai, 2022; Ngai et al., 2024; Haddad and Kattan, 2024). In a closely related work, Afridi et al. (2018) show that the LFP of married women in rural India decreased between 1987 and 2011, with increasing education levels among women (and men) as the most prominent factor. They suggest that changing returns to more educated women's work at home relative to market production helps explain the adverse effect on women's LFP in rural India. We complement Afridi et al. (2018) in two important ways. First, while the decline documented in rural India is from a baseline FLFP of around 50%, we establish a similar phenomenon in a context where the LFP of women is over 90%. Second, by examining the North-South differences in Vietnamese women's responses, we highlight the role of gender role attitudes in shaping the relationship between income and women's participation in market work.

We also add to the growing body of literature that highlights social norms as a key determinant of women's LFP and economic inclusion (see e.g., Fernández et al., 2004; Fortin, 2005; Fernández and Fogli, 2009; Maurin and Moschion, 2009; Doepke et al., 2012; Alesina

et al., 2013; Bertrand et al., 2015, 2021; Jayachandran, 2015; Bursztyn et al., 2017, 2020; Olivetti et al., 2020; Charles et al., 2022; Cortes et al., 2022; and Boelmann et al., 2024). In particular, we address studies exploring the impact of socialist legacies on women's labor market outcomes (see Bauernschuster and Rainer, 2012; Campa and Serafinelli, 2019; Fuchs-Schündeln and Schündeln, 2020; Lippmann et al., 2020; Boelmann et al., 2024). Our major departure from existing studies lies in our focus on the evolution of women's labor supply in response to positive economic shocks rather than static cross-sectional comparisons and in relating this to past political regimes and experiences during the Vietnam War.

In addition, we speak to the literature on the gendered implications of global integration. Cross-country studies show mixed results, varying by time period, income, and economic structure (Bussmann, 2009; Gaddis and Klasen, 2014). Using microdata, Gaddis and Pieters (2017) find no benefits for women from free trade in Brazil, while Juhn et al. (2013) show that technology advancements in Mexican exporting firms reduced gender wage and employment inequality. In China, Wang et al. (2022) find that import competition helped retain women in the labor force, whereas Keller and Utar (2022) report a shift toward family roles among Danish women, leading to long-term earnings losses. Erten and Keskin (2024) show that the differential labor market gains from Cambodia's WTO accession—favoring women over men—led to an increase in intimate partner violence. We contribute by examining export shocks in Vietnam, showing that women benefit more than men from trade openness, particularly in transitioning out of agriculture and into formal employment.

Furthermore, we relate to the literature examining the labor market impacts of Vietnam's trade liberalization and export-driven economic growth in recent decades. While prior studies have focused on key events such as the liberalization of rice trade in the 1990s (Edmonds and Pavcnik, 2005a, 2005b) or the 2001 US-Vietnam Bilateral Trade Agreement (McCaig, 2011; Fukase, 2013; McCaig and Pavcnik, 2015, 2018; Hoang and Nguyen, 2020),

our research centers on the subsequent two decades, from 2002 to 2018. By examining this more recent period, we can analyze the long-term effects of trade liberalization and exportdriven growth on the Vietnamese labor market.

The paper is organized as follows. Section 2 briefly summarizes Vietnam's sociopolitico landscape in recent history. Section 3 presents our empirical strategy, followed by data description in Section 4. Section 5 provides our main estimates, and Section 6 examines heterogeneity effects across regions. Section 7 presents further analysis based on the postwar migration policy as a natural experiment. Section 8 concludes.

2. Background

2.1 Colonial era and the division of Vietnam

Vietnam had traditionally been a patriarchal society with strong Confucian values (Ly 2021; Whitmore 2023), whose distinct emphasis on gender roles led to the subordination of women. When the Indochina peninsula came under French control in the second half of the 19th century, the French claimed authority over three regions of Vietnam: Tonkin in the north, Annam in the center, and Cochinchina in the south. Based on available data reconciled by Banens (2000), the population growth rates in Tonkin and Cochinchina before 1945 show striking similarities (see Table A2). At the end of the French colonial rule in Vietnam, the 1954 Geneva Agreements to resolve outstanding issues in French Indochina had resulted in Vietnam's partition into two zones, with the Democratic Republic of Vietnam (or the socialist North Vietnam) governing the north and the State of Vietnam (the future Republic of Vietnam, or the anti-communist South Vietnam) governing the south.⁵ The demarcation line at the 17th parallel split Tonkin and

⁵ The formation of the two states preceded the Geneva Agreements. The Democratic Republic of Vietnam was created in 1945, backed by the Soviet Union and China. The State of Vietnam was established in 1949, supported by France.

the northern half of Annam to the north, while Cochinchina and the southern half of Annam to the south. The separation was intended to be temporary, with a general election to be held by July 1956 to unify the country. Such an election was never realized, and the country remained divided over the next two decades until 1975.

2.2 Wartime, socialism, and the promotion of gender-egalitarian values in North Vietnam

During the period 1954-1975, the socialist government in North Vietnam advocated for a widespread improvement of Vietnamese women's social and political roles. The 1949 constitution stipulated that "women are equal to men in all respects" (Werner, 1981; Goodkind, 1995; Bryant, 1998). Land reforms and agricultural collectivization initiated in the early 1950s actively encouraged women's equal participation in collective labor, alongside the introduction of paid maternity leave—a feature entirely absent in anti-communist South Vietnam. Furthermore, the education level of Northern women had caught up with that of men for cohorts born after 1965, a milestone South Vietnam achieved only after reunification (see Figure A2).

Secondly, wartime necessity during the height of the Vietnam War (1965-1973) substantially increased Northern Vietnamese women's economic and political position. With the mass mobilization of men to the army, women assumed primary labor roles and, at the same time, engaged in civil activities. Their image was promoted as warriors and heroines of war (Werner and Belanger 2002).⁶ The Communist Party mandated gender quotas, aiming for women to occupy 30% of leadership positions by 1960 and increase female labor share to at least 35% of the workforce, particularly in light industries, education, and health care

⁶ The "Three Responsibilities" campaign called for women to contribute to war effort by replacing men enlisted in the army and taking up production, taking charge of family affairs and local administration, and encouraging their husband and sons to fight for reunification of the country.

(Resolution 153, 1967).⁷ These campaigns were accompanied by training programs for women, the promotion of female cadres, and the provision of public childcare in nearly half of production cooperatives (Mai and Le, 1978).

2.3 Economic and social changes in reunified Vietnam

The two regions reunified in 1976 with the end of the Vietnam War and following the fall of Saigon, and Vietnam has since remained a *socialist* republic. Agricultural collectivization was implemented in the southern provinces, and the whole country underwent a rigidly planned economy (Tri 1988; Beresford 1988). By the late 1980s, nationwide educational parity between genders was achieved (Figure A2), alongside initiatives such as equal pay and generous childcare provision to incentivize female workforce participation (Werner 1981).

However, with the massive economic reforms in the late 1980s that began shifting Vietnam towards a "socialist-oriented market economy under state guidance" (i.e., the "Doi Moi" reforms initiated in 1986), the country's politico-economic stance on the gender role has been altered. Men have become associated with technology, while there has been a renewed emphasis on the importance of family for women. At the same time, there have been severe cuts to parental leave and subsidized childcare (Goodkind 1995; Nguyen 1999). Despite being more prevalent in former North Vietnam, childcare supply experienced nationwide stagnation between the decade following "Doi Moi" (Figure A3), a trend worth noting as reductions in childcare supply could have important implications on female labor supply.

⁷ During 1961 to 1979, female ratio in the total workforce in North Vietnam increased from 43 to 63% in Light Industry, 12% to 60% in Education, 57% to 80% in Agriculture. On political and administrative spheres, the share of female public employees went from 5% in 1954 to 43% in 1979, share of female delegates in National Assembly went from 2% in 1946 to 30% in 1970. The percentage of female village leaders and female heads of production brigades was around 40% and 20%, respectively, in 1970 (Werner 1981).

Studies point to the post-socialist reversal of gender roles to the traditional male breadwinner model, as experienced in Eastern Europe and the former Soviet Union, as well as in China (LaFont 2001; Dumančić 2017). Nonetheless, Vietnam's FLFP rate has remained remarkably high at around 80% over the last three decades (International Labour Organization, 2022).⁸ Socialist ideology, prioritizing gender role equalization and recognizing the necessity of female labor for rapid economic growth and modernization, contributes to this progress. Additionally, the "missing men" effect, resulting from imbalanced sex ratios due to high military mobilization and male mortality during the Vietnam War, left many women as primary breadwinners. Research suggests that the former has a more enduring impact on women's labor market attachment, pointing out that living under the communist North has a larger effect compared to the "missing men" effect (Kreibaum and Klasen 2015). Consistent with this line of argument, Figure 2 shows higher LFP rate of Northern women, who behave very similarly to men, than for Southern women.

[Figure 2]

Data from the World Values Survey for 2002, 2006, and 2019 also suggest Southern Vietnamese hold more conservative gender role attitudes. more conservative gender role attitudes held by the Southern Vietnamese. The Northern and Southern Vietnamese, however, do not systematically differ in their beliefs on other, non-gender-related dimensions. We come back to this point in section 6.1.

Vietnam's reopening to the world economy and global integration has brought about high economic growth, making Vietnam "the dragon that rose from the ashes" (Tarp 2018). Since the 2000s, Vietnam has been experiencing inflows of FDI and rapid export expansion due in part to the enactment of the US-Vietnam Bilateral Trade Agreement (UVBTA) in 2001

⁸ International Labour Organization. "ILO Modelled Estimates and Projections database (ILOEST)" ILOSTAT. Accessed March 01, 2022. <u>https://ilostat.ilo.org/data/</u>.

and Vietnam's entry into the World Trade Organization (WTO) in 2007 (see Figure A1 and Table A1).

3. Empirical Strategy

Our goal is to understand how an exogenous expansion in economic opportunities between 2002-2018 affects women's labor market engagement, and how the pattern may differ according to the prevailing gender role attitudes. To proxy for the exogenous growth of the local economy, we use predicted exports for each province and year, based on a Bartik instrument. Specifically, we allocate nationwide industry-specific exports to provinces based on their initial industry composition prior to 2002, following the approach of Topalova (2010), Autor et al. (2013), and Kovak (2013).

Consider a variable $Exposure_{pt}$, which captures the (log of) predicted export per worker for province p in year t. We construct this variable by allocating nationwide industry-specific exports in each year to provinces according to their initial industry composition (based on industry-specific shares of local employment):

$$Exposure_{pt} = \log\left(\sum_{j} \frac{L_{pj,1999}}{L_{p,1999}} \times \frac{Export_{j,t}}{L_{j,1999}}\right)$$
(1)

where $L_{pj,1999}$ is province p's employment in industry j in 1999, and $L_{p,1999}$ is province p's total employment in 1999.⁹ The variable $Export_{jt}$ is Vietnam's nationwide total export value in industry j in year t, which we scale by nationwide initial total employment in industry j, $L_{j,1999}$, similar to Autor et al. (2013).

⁹ We use 1999 to measure sectoral and provincial employment in the initial (pre-2002) period, since the Vietnamese Census 1999 is the closest we can get to 2002, the beginning of our analysis window.

The exposure measure in equation (1) is a proxy of the local economy's development level, based on provincial exports driven by industry-specific global demand rather than unobserved local factors that may also correlate with women's labor market behavior. While data on actual exports at the province-by-year level are unavailable, we find that predicted exports by province and year show a significant positive correlation with gross regional domestic product (GRDP) per capita, which is available for the years 2010 to 2018 (see Figure 3). In section 5.1, we show that the results using this indicator for a subsample remain consistent with our baseline estimates.

[Figure 3]

We examine how predicted provincial export shocks—hereinafter "export exposure" or "export shock" for simplicity—affect individuals' labor market outcomes, by estimating the following difference-in-differences (DID) equation:

$$y_{ipt} = \alpha_0 + \alpha_1 Exposure_{pt} + \phi_p + \psi_t + X_{ipt}\lambda + Z_p\gamma_t + \epsilon_{ipt}$$
(2)

where y_{ipt} is the labor market outcome of individual *i* in province *p* in year *t*, such as working status. The variable *Exposure*_{pt} is the (log of) province-by-year level predicted exports, as constructed in equation (1). Throughout the analysis, we condition on both province fixed effects (FE) (ϕ_p) and year FE (ψ_t), accounting for province-specific and time-invariant unobservables as well as common nationwide shocks that vary over time, respectively. In addition, we control for individual-level characteristics with vector X_{ipt} , which includes an indicator for ethnic Viet (the majority ethnic group), gender, a cubic polynomial in age, and four categories of education (no education, primary, lower secondary, and upper secondary or higher).

Furthermore, we condition on a vector Z_p which includes a rich set of province-level initial (1999) characteristics, allowing for its effects to differ by time (captured in γ_t). In particular, we always include the initial employment share in manufacturing (interacted with

year FE) to account for the fact that export growth in Vietnam is primarily driven by the manufacturing sector (see Figure A1) and labor market trends in provinces with high versus low initial shares of manufacturing employment may systematically differ. By conditioning on the share of manufacturing industries in the baseline period (interacted with year FE), we use only the cross-sectional variation in the composition of industries within manufacturing (and within non-manufacturing) across regions for identification. Additionally, we control for the 1999 levels of gross output per capita in manufacturing and services, FLFP rates, and childcare supply (i.e., number of kindergarten classes per child aged 0-5), all interacting with year FE. These initial conditions further account for factors that may lead to differential evolution of female labor supply in different provinces. We cluster standard errors at the province level.

Our main coefficient of interest is α_1 , which captures the effect of export exposure on labor market outcomes. Our identifying assumption is that conditional on the included controls in equation (2), there is no correlation between $Exposure_{pt}$ and ϵ_{ipt} . In light of recent developments in the shift-share literature (see Goldsmith-Pinkham et al., 2020; Borusyak et al., 2022), we argue that the validity of our identifying assumption stems from "exogenous shares." In Appendix B, we present several diagnostic tests—suggested in Goldsmith-Pinkham et al. (2020) and Borusyak et al. (2025)—that support the plausibility of this assumption.

To uncover the differential effects of export expansion by gender, we consider a variant of equation (2), where we additionally include the interaction between $Exposure_{pt}$ and a female indicator, $Female_i$, while allowing for gender-specific year FE ψ_{gt} (instead of year FE ψ_t). Furthermore, we also investigate North-South differences, interacting $Exposure_{pt}$ with a dummy indicating the province being in South Vietnam.

4. Data

Our main data stems from the Vietnam Household Living Standard Surveys (VHLSS) covering the period 2002-2018. The surveys are conducted every two years by the General Statistics Office (GSO) of Vietnam, with technical assistance from the World Bank. The VHLSS is nationally representative and by far the most comprehensive microdata set in Vietnam on the living standards of the population. We restrict our attention to individuals of working age (20-64) in each survey year, resulting in a total sample size of 402,833. In Table 1, we provide the average characteristics of individuals in our sample for the period 2002-2018.

[Table 1]

Our main outcome of interest is "work," a dummy indicating whether the individual has worked at any time during the previous 12 months. We refer to this as an indicator of LFP. As Figure 2 shows, the participation rate is very high for both men and women in both North and South Vietnam. It is, however, always lowest among Southern women (around 85% in 2002, compared to 93% for the other groups—Northern men, Northern women, and Southern men), and trending downward for this group.

Based on industries, we classify individuals into agriculture, manufacturing, construction, or service sectors. Based on the type of employer, we can distinguish between the formal sector (state enterprises, collectives, domestic or foreign firms) and the informal sector (self-employed or household business). Figure A4 documents the rapid transformation of Vietnam's economic structure since the 2000s, with a visible shift from agriculture to manufacturing and services for both men and women.

To capture the initial industry composition in 60 provinces prior to Vietnam's trade liberalization in the 2000s, we use the Vietnam Population and Housing Census 1999 (or the Census 1999), which consists of 3% of the population with 2,368,167 individuals, of which 1,135,981 are in the labor force.¹⁰ Workers are employed in 153 industries according to International Standard Industrial Classification of All Economic Activities Revision 3 (ISIC3) at 3-digit level.

To obtain Vietnam's exports by industry, we use the United Nations Comtrade database (UN Comtrade) for the period 2002-2018. UN Comtrade provides detailed trade data at the product level using the Harmonized Coding and Description System (HS). We convert products (HS 6-digit level) to industries (ISIC3 3-digit level), using the concordance tables provided by the World Integrated Trade Solution.¹¹ Export values are expressed in 2018 VND based on annual VND/USD exchange rates and the Consumer Price Index obtained from Vietnam's General Statistics Office (GSO).

In equation (1), we construct the variable $Exposure_{pt}$ by combining each province's industry-specific employment shares in 1999 with nationwide export values per worker at the industry-by-year level. The mean (SD) of $Exposure_{pt}$ is 17.271 (0.952), see Table A1. The table also lists the top 10 exporting provinces in 2002 and 2018, together with the top 10 provinces experiencing the largest changes in $Exposure_{pt}$ over the study period.

As a first check, Figure 3 plots the correlation between export exposure and log GRDP per capita, available for 2010-2018.¹² The estimates come from the regression of log GRDP per capita at the province-year level on export exposure, controlling for a province's employment share in manufacturing in 1999. The regression is weighted by provinces' initial population size. As expected, our predicted export measure $Exposure_{pt}$ is highly correlated with the province's actual development indicator.

¹⁰ In 2002, Vietnam had 61 provinces. In 2003, three new provinces were created by splitting Dien Bien from Lai Chau, Dak Nong from Dak Lak, and Hau Giang from Can Tho. In 2008, Ha Tay was merged into Ha Noi, resulting in a total number of 63 provinces. Taking these changes into account, we define 60 provinces with consistent geographical boundaries over the study period.

¹¹ <u>https://wits.worldbank.org/product_concordance.html</u>.

¹² Data on GRDP per capita by province are provided by the General Statistics Office (GSO) of Vietnam, expressed in constant 2018 VND.

In Figure 4, we illustrate the spatial variation in $Exposure_{pt}$. Panel (a) shows the levels of $Exposure_{pt}$ for 2002 and 2018, respectively, while Panel (b) shows the change in $Exposure_{pt}$ between 2002 and 2018. These measures vary widely across Vietnam, as well as within both the North and South.

[Figure 4]

5. Export-induced Economic Development and Female Labor Supply

5.1 Overall effects

We start by documenting the effects of export-driven economic expansion on the LFP of Vietnamese men and women over the period 2002-2018. Table 2 presents the estimates of equation (2) using the working status of individuals as the dependent variable. The variable "work" indicates whether the person ever worked in the last 12 months. The first two columns show the average effects for all individuals, while the last two columns report the differential effects of export exposure by gender. Individual-level controls include ethnicity, gender, a cubic polynomial in age, and education level. All specifications (except column 4) include an interaction between the initial share of employment in manufacturing with year FE, so that we only exploit variation in sub-industries within manufacturing and within non-manufacturing in the base period (1999).

[Table 2]

Column 1 shows a negative, albeit insignificant, effect of export exposure on work status. Column 2 controls for the full set of initial province social and economic conditions interacted with year FE, aiming to address potential confounding factors. Specifically, we control for a province's initial gross output per capita in manufacturing and service. This accounts for the possibility that FLFP in provinces at varying levels of economic development follow different trajectories, confounding the effect of export exposure on FLFP. We also control the initial provincial FLFP and childcare supply (proxied by the number of kindergarten classes per 0-5-year-old child). Given that childcare availability is an important determinant of women's decision to supply market work, controlling for childcare supply rules out this potential confounder. We find that an increase in export exposure has a small but negative effect on overall LFP.

This result, however, masks substantial gender differences. Column 3 separates the effects by gender, using a triple difference approach and allowing for gender-specific year FE. In provinces with one SD (0.952) higher export exposure, equivalent to export growth of 160%, women's working status declines by 2.7 pp ((-0.031 + 0.003) * 0.952), while there is no corresponding effect for men. This translates to a 3 pp widening of the gender gap in LFP, or 56% of the mean gender gap (0.03 / (0.918 - 0.865)).¹³ This gender differential effect remains virtually unchanged in column 4, where we control for province-by-year FE, allowing provinces to follow different trends.¹⁴

Alternative measures of economic growth. In Table 3, we gauge the robustness of our findings with alternative measures of provincial economic development. Columns 1 and 2 replicate our baseline estimates from columns 2 and 3 of Table 2, where export exposure is constructed using all traded industries. In columns 3 and 4, we construct export exposure based on manufacturing industries only. The estimates remain largely unchanged. This is reassuring given that manufacturing is the primary driver of exports in Vietnam.

¹³ Wang et al. (2022), examining the effect of import competition in China, find that in a prefecture experienced average tariff reduction (30 pp), the gender employment gap is reduced by about 6 pp.

¹⁴ Note that in column 4, the province-by-year FE absorbs the base exposure variable, which varies at the provinceyear level. Moreover, the interactions between province-specific initial characteristics and year FE are also absorbed by the province-by-year FE.

[Table 3]

In columns 5 to 8, we use a subsample that restricts the time window to 2010-2018, during which data on provincial GRDP per capita are available. Columns 5 and 6 repeat the analysis using our main export exposure indicator—using all traded industries, showing that the effects are also robust to the choice of time frame. The estimates in column 6 show that in response to a one SD increase in export exposure, women's LFP decreases by 2.2 pp, or 2.5 percent of the mean (0.022 / 0.859). In columns 7 and 8, we use log GRDP per capita as an explanatory variable. A one SD increase in log GRDP per capita (0.518) is associated with a 2 pp (0.518 * (-0.030 - 0.009)) reduction in FLFP, or 2.3 percent of the mean (0.02 / 0.859). Although less precisely estimated, the results obtained using GRDP per capita are consistent with our baseline, demonstrating that using export exposure as a proxy for economic development is reasonable.

Export shocks and sectoral allocation of labor. One might suspect that the observed decrease in FLFP documented above is due to the displacement of female labor during trade liberalization. To shed light on this, we examine the effects of export exposure on economic sector and employer type, conditional on working. Results are reported in Table 4, using our preferred specification from column 2 (for Panel A) and column 3 (for Table B) of Table 2.

[Table 4]

Panel A shows that export shocks shift workers out of agriculture (column 1) and into manufacturing and services (columns 2 and 4), based on the sample of all individuals who have worked in the previous 12 months. Moreover, the positive export shocks in the local economy facilitate the transition of the workforce to the formal sector, particularly foreign-owned firms (columns 5 and 6), consistent with prior findings on trade-induced formalization in Vietnam

within the manufacturing sector (see McCaig and Pacvnik, 2018).¹⁵ Panel B reveals that, when remaining in the workforce, female workers benefit more than male workers from these emerging economic opportunities, with a swifter transition out of agriculture and into manufacturing and services as well as foreign firms. Specifically, a one SD increase in export exposure (0.952) disproportionately raises women's probability of employment (relative to men) in manufacturing by 1.4 pp (9.3% of the female average) and in services by 1.8 pp (5.5% of the female average). This trend aligns with the rise of FDI in low-skilled, labor-intensive manufacturing in Vietnam (McCaig, 2011).

In addition, we examine the effect of export exposure on wages. Specifically, we repeat the analysis from Table 2, replacing the dependent variable with log annual earnings (expressed as 1,000 VND in 2018).¹⁶ If the decline in the employment rate is due to a negative labor demand shock, then we would expect a negative relationship between export exposure and wages. In contrast, if the decline in the employment rate is supply-driven, we would expect a positive (or nonnegative) relationship between export exposure and wages and export exposure are positively correlated. This holds for both men and women, with no significant gender difference. The results remain consistent whether using earnings from the main job (columns 1 and 2) or from all jobs combined (columns 3 and 4).¹⁷ Overall, the evidence suggests that the decline in women's participation is unlikely due to adverse labor demand shocks disproportionately affecting female workers.

¹⁵ Formal firms include all state enterprises, collectives, foreign firms, and registered private businesses under the Enterprise Law. Self-employed individuals or household businesses, typically with fewer than two workers and unregistered, are classified as informal.

¹⁶ In the VHLSS data, we observe wages for a subsample of individuals who reported receiving salaries in the past 12 months. This subsample comprises around 34% (42%) of the full sample for annual earnings from the main job (from all jobs).

¹⁷ In Table A4, we zoom in on North and South Vietnam. For ease of interpretation, the coefficients in this table are reported as the total effect for each gender-by-region group, rather than using men as the reference group as in the main analysis. The positive relationship between wages and export exposure is also observed for both men and women in each region. Point estimates for wage increases are largest for Southern women, although the difference compared to Northern women is not statistically significant. This result holds across all specifications, whether using annual earnings from the main job only versus all jobs combined (columns 1 and 4), or when excluding the southernmost provinces whose institutions differ from the rest of Vietnam (columns 2, 3, 5, and 6).

5.2 Heterogeneity by marital status and household income

Why would Vietnamese women reduce their labor supply amid expanding economic opportunities? To further understand the drivers behind this phenomenon, we investigate possible heterogeneity in women's responses based on household characteristics. In Table 5, we examine whether the labor supply response to export exposure differs by marital status.¹⁸ Column 1 reports separate estimates for single and married individuals. Column 2 further reports the differential effects by gender, comparing single women with single men, and married women with married men. The results indicate that the negative effect of export exposure on employment is concentrated among married individuals, particularly married women, while there is no corresponding effect for married men. In contrast, the impact of export exposure on singles is positive, with a small and statistically insignificant gender difference.

[Table 5]

Restricting our attention to married women, we further examine heterogeneity by household wealth and husband characteristics in Table 6. In each column, we use all observations that can be matched to the corresponding household or husband characteristics. The number of observations in columns 2 and 3 is smaller since wage and earnings information is available only for those in wage employment.¹⁹ Based on columns 1-3, we find that relative to the reference group (lowest quartile), the reduction in the working rate is stronger for women in the highest quartile of household wealth or husbands' wages/earnings. In column 4, we find no significant difference across husbands' education levels. We return to this point in the next section, where we highlight substantial regional heterogeneity between North and South

¹⁸ The share of married individuals in our sample is 84% (see Table 1).

¹⁹ Nearly all individuals in the formal sector report wage and earnings, but only around one in every five informal workers reports this information. The results remain virtually unchanged when excluding women with husbands in agriculture from the sample.

Vietnam. In column 5, we find that women with husbands in medium or high-skilled occupations exhibit a stronger decrease in labor supply, compared to the reference group in low-skilled occupations.²⁰

[Table 6]

The concentration of negative effects on the LFP among married women—particularly those in wealthier households or those with higher-skilled husbands—indicates that householdlevel labor allocation between home and market work matters. With the trade liberalization in the 2000s and the subsequent rapid rise in income, Vietnamese families may have found that it was no longer necessary for both spouses to work. If the husband's income is sufficient, women may reduce their labor supply to enjoy more leisure and/or specialize in home production. This suggests that whether the income effect (i.e., higher wages inducing women to drop out of the labor market) dominates the substitution effect (i.e., higher wages prompting women to increase their labor supply) in the context of export-driven economic growth may depend on the prevailing gender role attitudes. We examine this hypothesis in the next section.

6. Labor market behavior of women in North and South Vietnam

To explore the hypothesis on how household-level decisions on labor allocation might be shaped by gender role attitudes, we examine potential heterogeneity in women's responses between South Vietnam (formerly capitalist) and North Vietnam (always socialist). The latter's sustained exposure to socialist ideologies—particularly gender-egalitarian values—may make them more resistant to the resurgence of the male breadwinner norm. This analysis is further

²⁰ We code the International Classification of Occupations (ISCO) occupation groups 1-2 as "Manager/Professional", groups 3-8 as "Medium-skilled", and group 9 as "Elementary". Armed forces occupations, representing 0.3% of total workforce, are excluded from the analysis. See <u>https://ilostat.ilo.org/methods/concepts-and-definitions/classification-occupation</u>.

motivated by the stylized facts regarding the differences in gender role attitudes between the two regions, as detailed below.

6.1 Survey evidence on gender role attitudes

To gain insights into regional variations in gender role attitudes in Vietnam, we draw on the World Values Survey (WVS). In Vietnam, the WVS was carried out in 2001 (wave 4), 2006 (wave 5), and 2019 (wave 7), allowing us to observe the attitudes of the Vietnamese people in the two regions on gender-related and gender-neutral issues. We pool all years and report the mean responses of the South Vietnamese (relative to the North Vietnamese, the reference group) with confidence intervals at 95% level, while conditioning on year FE.

In the first panel of Figure 5, we report the North-South difference in gender role attitudes, focusing on respondents' (dis)agreement with statements such as, "When jobs are scarce, men should have more rights to a job than women," "University is more important for a boy than for a girl," "Men make better business executives than women do," and "[There is a] problem if women have more income than [their] husbands."²¹ As shown, respondents from the South exhibit more conservative attitudes than their Northern counterparts, and this is true for both males and females (see Figure A5). Conversely, there is no systematic difference between the two regions regarding core beliefs on gender-neutral dimensions such as effort, luck, and fairness (as seen in the second panel of Figure 5). Overall, evidence from the survey data aligns with the enduring disparities in gender role attitudes between Northern and Southern Vietnamese populations, stemming from the country's prior division and wartime experiences.

[Figure 5]

²¹ We recode "Strongly agree," "Agree," and neutral responses as 1 (Agree), while "Disagree" and "Strongly disagree" are recoded as 0 (Disagree).

It is worth noticing that the historical state of Vietnam, the Dai Viet kingdom, was heavily influenced by Confucianism, which has long-lasting negative impacts on gender equality (Ly, 2021; Vu and Yamada, 2023; Whitmore, 2023). With the kingdom's gradual expansion from northern Vietnam to the southern regions over a millennium, one might expect more entrenched traditional gender roles in the North compared to the South. While historical data during the French colonial period (before 1945) is scarce, early 20th-century vital statistics support this view. Banens (2000), reconciling from various sources, notes serious underregistration of female births all over the country.²² In Cochinchina (southern Vietnam) during 1915-1937, the male/female ratio at birth ranged from 1.16 to 1.19. In Tonkin (northern Vietnam), the ratio was even higher—1.12 in hospitals and 1.71 for home births (Gourou, 1936; Banens, 2000)—implying a stronger son preference and higher gender inequality in the North.²³ However, the prevailing higher gender-egalitarian values in North Vietnam War on gender role attitudes.

6.2 Heterogeneous effects on FLFP across regions

In Table 7, we examine the differential effects of export shocks on working rate by gender and region (North vs. South), in a variant of equation (2). Here, we allow for gender-by-region FE, in addition to all the included controls in column 3 of Table 2. Section 2 describes the North-South boundary at the 17th parallel north (see panel (b) of Figure 4). Columns 1-3 are based on the full sample while columns 4-6 restrict attention to the sample of married women.

 $^{^{22}}$ According to Banens (2000), the male/female ratio at birth has been normal (1.05) over the reconstituted period of the 20th century, with similar population growth rates across regions (see Table A2).

²³ This is a reasonable assumption given Tonkin's proximity to China and, as such, is more influenced by Confucianism. While the Viet people have inhabited the northern part of Vietnam for thousands of years, their settlement in the southern part started relatively recently, just three centuries ago.

Column 1 shows that while men's participation is hardly affected, women's participation in the labor market significantly decreases in both regions with an increase in export exposure. Looking at $\beta_1 - \alpha_1$ (which reflects the South-North difference in women's disproportionate response relative to men), we find that the decline in women's participation is roughly 1 pp larger in the South than in the North for one SD increase in export exposure.

[Table 7]

North-South historical differences. To ensure that the North-South difference in FLFP responses stems from the different political regimes and experiences during the Vietnam War (1954-1975) rather than other cultural and institutional differences between regions, we consider subsamples constituting the various territories of Vietnam, reflecting the historical and geographical development of the country.

The southernmost part of Vietnam (consisting of 12 provinces in the Mekong River Delta and three other southeastern provinces) was not incorporated into Vietnam until 1833.²⁴ The region once belonged to the Khmer Empire, potentially subjecting it to very different institutions and norms from the rest of Vietnam (historically Dai Viet). Using a regression discontinuity design, Dell et al. (2018) show persistent differences for villages on the two sides of the Dai Viet – Khmer border. In particular, economic development and living standards on the Dai Viet side of the boundary are higher, due to stronger norms regarding local cooperation and civic engagement. In addition, the Mekong River Delta plays a vital role in the agricultural production and food security of Vietnam as a whole: the delta produces half of the country's rice and one-third of its GDP, factors that may interact with female employment if women are

²⁴ The country gradually expanded southward (with the "Nam tien" movement, literally translated into moving towards the south), conquering the Kingdom of Champa along the central coast of Vietnam (starting from the beginning of the 14th century and completed in the end of the 17th century). Shortly afterward, Vietnamese settlers established the province of Gia Dinh (now Ho Chi Minh City).

disproportionately employed in agriculture. For these reasons, we first drop from the sample the 12 provinces in the Mekong River Delta (column 2) and then all the 15 that were historically on the Khmer side (column 3).

Upon excluding the southernmost provinces, point estimates suggest that the reduction in FLFP is more than twice as large in the South, and the difference is significant at the 1% level. This reinforces our hypothesis of cultural divergence between the two regions during the war. In columns 4-6, we repeat the analysis for married individuals only. The point estimates are somewhat larger but remain consistent with the results from the full sample.

North-South differences in economic structure. One could argue that the stronger reduction of female labor supply in the South of Vietnam is due to different economic structures between the two regions that may create relatively fewer employment opportunities for Southern women. One may also believe that due to the larger agricultural sector in the South, Southern women might face greater displacement when agricultural jobs disappear. Data from the Census 1999, however, do not support these conjectures. Panel (a) of Figure A6 illustrates that, in 1999, the economic structure of the two regions was similar.²⁵ Zooming into the manufacturing sector (Panel (b)), we see that although female and male workers concentrate in different industries, there is little North-South difference within gender.

In Table 8, we conduct a more direct check to support the argument that regional economic structures do not explain the stronger reduction in FLFP in the South, focusing on married individuals. To isolate the North-South behavioral differences, we estimate the effects of provincial export exposure predicted separately for female-intensive and male-intensive industries (see equation (1)). This approach ensures that the North-South difference in FLFP is

²⁵ If anything, a larger share of the workforce in the North is engaged in agricultural activities and slightly less engaged in manufacturing.

not driven by variations in the composition of exporting industries, particularly the femalefriendliness of the jobs created. We define an industry as female-intensive if the share of female workers in that industry in 1999 is above the median (around 36%).

[Table 8]

For ease of comparison, column 1 replicates the baseline results for the married sample from column 4 of Table 7. In columns 2 and 3, we observe negative effects on women's working status in response to either method of constructing the provincial export exposure both of which are strongly correlated, with the effects being larger in the South. Columns 4-6 confirm that these patterns hold when using only manufacturing industries to construct provincial export exposure.

North-South differences in home responsibilities. A remaining concern is that the observed North–South differences may stem from differing demands for women's home production across the two regions. For example, women in the South may have more children or larger families to care for, which could make it more difficult for them to participate in market work.

Using VHLSS data, Figure A7 shows similar household sizes in North and South Vietnam, with a gradual decline from 2002 to 2018. Using census data from 1999, 2009, and 2019, Figure A8 plots births per woman by age group (15–24, 25–39, 40–49), revealing similar declining trends across regions. The declining trend in fertility is most substantial for women aged 40–49, which aligns with the end of their fertile age and more closely reflects the total fertility rate. This pattern alleviates the concern that greater home responsibilities, rather than traditional gender role attitudes, are the primary drivers of North-South differences in women's participation in market work.

6.3 Heterogeneity by household income

In the previous section, we establish that the stronger trade-induced reduction FLFP of Southern compared to Northern women can be attributed to enduring differences in gender norms between the two regions, rather than other economic structure or institutional factors. How do these differences in gender norms influence the response of women across the wealth and income distributions? In this section, we revisit the heterogeneity analysis by household and husband's characteristics in Table 6, allowing for differential effects between the North and South. Results are presented in Figures 6 and 7 (estimates are reported in Tables A5 and A6, respectively).

[Figure 6]

[Figure 7]

While the negative impact on female labor supply in the North is primarily observed among women married to husbands in the top earnings quartile or in wealthier households, this gradient is absent in the South, where women reduce their labor supply across all quartiles of household wealth and husbands' earnings (Figure 6). This suggests that women's labor market attachment is weaker in the South, prompting them to decrease market work as soon as it becomes financially feasible. In contrast, Northern women demonstrate stronger labor market attachment, with the income effect becoming dominant only at a higher household income threshold. A similar pattern emerges when considering the husband's education and skill levels (Figure 7). These results highlight the interplay between gender norms and socioeconomic status: in traditional male breadwinner societies, the income effect manifests at lower thresholds, while in gender-egalitarian societies, it dominates at higher income levels. These results remain robust when excluding 12 provinces in the Mekong River Delta or all 15 provinces from the former Khmer Kingdom.

7. Further Evidence from Postwar Migration Policy

7.1 New Economic Zones program and North-born population in the South

We have so far argued that gender role attitudes, nurtured under different political regimes and experiences during the Vietnam War era are driving the North-South differences in women's labor supply responses to trade-induced regional economic growth. Still, one could argue that North and South Vietnam have always been vastly diverse geographically, economically, and culturally. To ensure that our results are indeed driven by different gender role attitudes prevalent in North vs. South Vietnam, rather than other deep-rooted institutional and environmental differences between the two regions, we exploit a historical event in which millions of people were moved across the country in the aftermath of the reunification of Vietnam.

Known as the New Economic Zones (NEZ) program and implemented by the Vietnam Communist Party, millions of people were relocated to uninhabited mountainous or rural areas in the southern and central regions, formerly part of South Vietnam. The NEZ policy played a vital role in Vietnamese planning, ranking as the top priority among all government agricultural projects. To ensure the success of these initiatives, the State integrated them into the five-year plans and annual socio-economic development plans (Jones 1982; Desbarats 1987; Dang et al. 1997; Gendreau 2000). During the first five-year period (1976-1980), over 1.5 million migrants (out of a target of four million) were relocated, 40% of whom came from the North. By the year 2000, more than five million people had relocated across the country. Table A7 provides details on planned and actual migration figures from 1976 to 2000.

The establishment of the NEZs served multiple objectives, including redistributing the population to alleviate overcrowding in the Red River Delta in the north and reducing urban population density in southern cities. Additionally, it aimed to reclaim lands for food

production to support a postwar nation facing food shortages, and address both internal and external security (General Statistics Office 1991).²⁶ Before the onset of Doi Moi in 1986, Vietnam's economy was heavily centralized, giving the Government substantial control over migratory flows. The organized resettlement program left migrants (or "resettlers") with little to no agency in determining their relocation. With the liberalization of the economy and the relaxation of social controls in recent years, internal migration has become increasingly less restricted.

The program's nature allows us to use it as a natural experiment, wherein North-born immigrants (with early exposure to socialism) are subject to the same environment as the Southern population (with later exposure to socialism).²⁷ Data from VHLSS 2014, 2016 and 2018 provide information on the birth province of individuals. Based on this information, we construct a dummy for birth origin, *BirthNorth*_i, indicating whether or not a person was born in North Vietnam. Figure 8 illustrates the shares of North-born individuals and all immigrants by provinces in South Vietnam. The eight darkest-shaded provinces are those with the highest share of immigrants, closely matching the designated New Economic Zones in the mountainous or rural areas. We refer to these provinces as the "Destination" ones.

[Figure 8]

In Figure A9, we present the average LFP rate by gender and place of birth (Southernborn vs. Northern-born) among individuals aged 20-64 residing in South Vietnam. The LFP rate among Northern-born women (living in the South) is very high, reaching as much as 90%

²⁶ The Central Highlands lie on the borders of Vietnam. The resettlement of inhabitants into thinly populated yet strategically sensitive areas near the Cambodian, Laotian, and Chinese borders was considered imperative for national defense (Desbarats 1987).

²⁷ In the context of North-South movement of employees at an Italian bank, Ichino and Maggi (2000) disentangle the effects of birth origin vs. current work location. In that paper, a small number of movers are assigned to a new environment in which the existing locals form the majority. In our context, the forced migrants from the North were placed in destinations that were largely uninhabited initially, making the environment conducive to retaining norms and values cultivated in the North.

in 2014. Although it declines slightly over time, their participation rate remains much higher than that of native Southern women.

7.2 Patterns of response by birth origin

In a sample of Southern provinces only, we estimate a variant of equation (2), interacting $Exposure_{pt}$ with a birth origin indicator (Northern-born or Southern-born) and with a gender indicator. We retain the sample of individuals who were born between 1954 and 1979, corresponding to the war and the early stage of the NEZ resettlement, thereby increasing the likelihood that they moved due to the program rather than self-selection into migration.²⁸ We additionally control for an interaction term between *BirthNorth*_i and *Destination*_p since Northern-born migrants living in the eight Destination provinces might have unobserved characteristics (e.g., reason for moving or relocation challenges) very different from those of people in the other areas, where *Destination*_p is an indicator for being one of the Destination provinces (see Figure 8).

Results are reported in Table 9. In column 1, the sample consists of individuals residing in all Southern provinces. In column 2, the sample is further restricted to migrants living in the South, including both Southern-born migrants and Northern-born migrants. Since the Northern-born migrants were born and grew up under the socialist North, their exposure to the socialist regime is almost 30 years longer than that of the Southerners. To the extent that former

 $^{^{28}}$ We exclude those born before 1954 due to the historical event where over 1 million Northern residents fled to the South during the 300-day Passage of Freedom following the 1954 Geneva Accords. This group, known as *Bac-54* (literally North-1954), predominantly comprised political and Catholic refugees fleeing potential persecution by the communist government. Second, we exclude cohorts born after 1979 since we do not observe the timing of migration and therefore cannot be sure that they moved under the NEZ policy, particularly since 1986. Additionally, the income effect is less relevant for younger individuals, who are also more likely to be single. As discussed in section 5, the reduction in FLFP is primarily observed among married women. Nevertheless, we find similar results when extending the sample to cohorts born 1954-1989, as reported in Table A8.

exposure to socialism has long-lasting effects on gender role attitudes, we expect to see the Northern-born population living in the South behave more similarly to people living in the North. This is indeed the case for women born between 1954 and 1979. Focusing on column 1, the effect of export exposure on women's LFP (relative to men) is -0.038 for the Southern-born. The corresponding estimate for the Northern-born women is -0.024, or 37% smaller in absolute terms compared to the Southern-born women. This suggests that while both Northern-born migrants and native Southern women reduce their labor supply in response to export shocks, Northern-born women do so to a lesser extent, despite sharing the same contemporaneous institutions.

In column 2, we obtain similar results when restricting the sample to migrants only. The differential effect for Northern-born women relative to Southern-born women ($\beta_1 - \alpha_1$) is -0.010, significant at the 1% level. Directly comparing migrants of different birth origins helps alleviate concerns about potential unobserved differences between migrants and non-migrants, safeguarding the finding on the North-South differential effect among women against inherent migration characteristics. Interestingly, the magnitude of this difference mirrors that of our benchmark result in Table 7 (column 1) when comparing women who currently live in North and South Vietnam. In columns 3 and 4 where we further restrict the sample to Destination provinces, the results on the relative reduction of FLFP by gender and birth origin hold.

[Table 9]

8. Conclusions

We explore the impact of gender role attitudes on women's labor supply in response to rising incomes and expanding economic opportunities. Our study focuses on reunified Vietnam, particularly during the trade liberalization and rapid growth of the 2000s. While Vietnam today

shares similar institutions, its history of division (1954-1975) resulted in starkly different regimes and experiences in the North and South. The socialist North promoted gender equality and women's economic participation, whereas the capitalist South maintained a traditional male breadwinner norm.

Utilizing quasi-random variation in provincial export growth from 2002 to 2018, we find that positive export shocks, while facilitating the transition of workers from agriculture to the formal sector, significantly reduce women's LFP—especially among married women whose husbands are in skilled occupations. This decline is more pronounced in the South than in the North, and among native Southerners compared to Northerners who migrated south after the war.

Our findings suggest that the prevailing gender role attitudes—whether the male breadwinner norm or gender-egalitarian values—can influence women's engagement in the labor market in contrasting ways as incomes rise. This has important implications for emerging economies. A key question for future research will be whether women in the North will resist the resurgence of the male breadwinner norm or gradually adopt behaviors similar to those of their Southern counterparts in the coming decades.
References

- Afridi, Farzana, Taryn Dinkelman, and Kanika Mahajan. 2018. "Why Are Fewer Married Women Joining the Work Force in Rural India? A Decomposition Analysis over Two Decades." *Journal of Population Economics* 31: 783-818.
- Alesina, Alberto, Paola Giuliano, and Nathan Nunn. 2013. "On the Origins of Gender Roles: Women and the Plough." *Quarterly Journal of Economics* 128 (2): 469–530.
- Antecol, Heather. 2000. "An Examination of Cross-Country Differences in the Gender Gap in Labor Force Participation Rates." *Labour Economics* 7 (4): 409–26.
- Autor, David H., David Dorn, and Gordon H. Hanson. 2013. "The China Syndrome: Local Labor Market Effects of Import Competition in the United States." *American Economic Review* 103 (6): 2121–68.
- Banens, Maks. 2000. "Vietnam: A Reconstitution of Its 20th Centuty Population History." In *Quantitative Economic History of Vietnam, 1900-1990: An International Workshop*, edited by Jean-Pascal Bassino, Jean-Dominique Giacometti, Kōnosuke Odaka, and Suzanne Ruth Clark. Tokyo: Institute of Economic Research, Hitotsubashi University.
- Bauernschuster, Stefan, and Helmut Rainer. 2012. "Political Regimes and the Family: How Sex-Role Attitudes Continue to Differ in Reunified Germany." *Journal of Population Economics* 25 (1): 5–27.
- Beresford, Melanie. 1988. "Issues in Economic Unification: Overcoming the Legacy of Separation."
 In *Postwar Vietnam: Dilemmas in Socialist Development*, edited by David G. Marr and Christine P. White, 95–110. Cornell University Press.
- Bertrand, Marianne, Emir Kamenica, and Jessica Pan. 2015. "Gender Identity and Relative Income within Households." *Quarterly Journal of Economics* 130 (2): 571–614.
- Bertrand, Marianne, Patricia Cortes, Claudia Olivetti, and Jessica Pan. 2021. "Social Norms, Labour Market Opportunities, and the Marriage Gap between Skilled and Unskilled

Women." Review of Economic Studies 88 (4): 1936–1978.

- Blau, Francine D., and Lawrence M. Kahn. 2013. "Female Labor Supply: Why is the United States Falling Behind?." *American Economic Review* 103 (3): 251-256.
- Blau, Francine D., Lawrence M. Kahn, and Kerry L. Papps. 2011. "Gender, Source Country Characteristics, and Labor Market Assimilation among Immigrants." *Review of Economics and Statistics* 93 (1): 43-58
- Boelmann, Barbara, Anna Raute, and Uta Schonberg. 2024. "Wind of Change? Cultural Determinants of Maternal Labor Supply." *American Economic Journal: Applied Economics*, forthcoming.
- Borusyak, Kirill, Peter Hull, and Xavier Jaravel. 2022. "Quasi-Experimental Shift-Share Research Designs." *The Review of Economic Studies* 89 (1): 181–213
- Borusyak, Kirill, Peter Hull, and Xavier Jaravel. 2025. "A Practical Guide to Shift-Share Instruments." *Journal of Economic Perspectives* 39 (1): 181–204.
- Boserup, Ester (1970). Woman's Role in Economic Development. St. Martin's Press.
- Bryant, John. 1998. "Communism, Poverty, and Demographic Change in North Vietnam." *Population and Development Review* 24 (2): 235–69.

Bulletin economique de l'Indochine. 1938. Hanoi.

- Bussmann, M. 2009. The Effect of Trade Openness on Women's Welfare and Work Life. World Development, 37 (6): 1027–1038.
- Bursztyn, Leonardo, Alessandra L. González, and David Yanagizawa-Drott. 2020. "Misperceived Social Norms: Women Working Outside the Home in Saudi Arabia." *American Economic Review* 110 (10): 2997–3029.
- Bursztyn, Leonardo, Thomas Fujiwara, and Amanda Pallais. 2017. "'Acting Wife': Marriage Market Incentives and Labor Market Investments." *American Economic Review* 107 (11): 3288–3319.

- Campa, Pamela, and Michel Serafinelli. 2019. "Politico-Economic Regimes and Attitudes: Female Workers under State Socialism." *Review of Economics and Statistics* 101 (2): 233–48.
- Charles, Kerwin Kofi, Jonathan Guryan, and Jessica Pan. "The Efects of Sexism on American Women: The Role of Norms vs. Discrimination." *Journal of Human Resources* (2022).
- Cortes, Patricia, Gizem Koşar, Jessica Pan, and Basit Zafar. 2022. "Should Mothers Work? How Perceptions of the Social Norm Affect Individual Attitudes Toward Work in the U.S." *NBER Working Paper 30606*.
- Dang, Anh, Sidney Goldstein, and James McNally. 1997. "Internal Migration and Development in Vietnam." *International Migration Review* 31 (2): 312–37.
- Dell, Melissa, Nathan Lane, and Pablo Querubin. 2018. "The Historical State, Local Collective Action, and Economic Development in Vietnam." *Econometrica* 86 (6): 2083–2121.
- Desbarats, Jacqueline. 1987. "Population Redistribution in the Socialist Republic of Vietnam." *Population and Development Review* 13 (1): 43–76.
- Dinkelman, Taryn, and L. Rachel Ngai. 2022. "Time Use and Gender in Africa in Times of Structural Transformation." *Journal of Economic Perspectives* 36 (1): 57-80.
- Doepke, Matthias, Michele Tertilt, and Alessandra Voena. 2012. "The Economics and Politics of Women's Rights." *Annual Review of Economics* 4 (1): 339-372.
- Dumančić, Marko. 2017. "The Communist and Postsocialist Gender Order in Russia and China." Chapter. In *The Cambridge History of Communism*, edited by Juliane Fürst, Silvio Pons, and Mark Selden, 3: 422–46. Cambridge: Cambridge University Press.
- Edmonds, Eric V., and Nina Pavcnik. 2005a. "Child Labor in the Global Economy." *Journal* of Economic Perspectives 19 (1): 199-220.
- Edmonds, Eric V., and Nina Pavcnik. 2005b. "The Effect of Trade Liberalization on Child Labor." *Journal of International Economics* 65 (2): 401-419.

- Erten, Bilge, and Pinar Keskin. 2024. "Trade-offs? The Impact of WTO Accession on Intimate Partner Violence in Cambodia." *Review of Economics and Statistics* 106 (2): 322-333.
- Fernández, Raquel, Alessandra Fogli, and Claudia Olivetti. 2004. "Mothers and Sons: Preference Formation and Female Labor Force Dynamics." *Quarterly Journal of Economics* 119 (4): 1249-1299.
- Fernández, Raquel, and Alessandra Fogli. 2009. "Culture: An Empirical Investigation of Beliefs, Work, and Fertility." American Economic Journal: Macroeconomics 1 (1): 146–77.
- Fisman, Raymond, and Edward Miguel. 2007. "Corruption, Norms, and Legal Enforcement: Evidence from Diplomatic Parking Tickets." *Journal of Political Economy* 115 (6): 1020– 1048.
- Fortin, Nicole M. 2005. "Gender Role Attitudes and the Labour-Market Outcomes of Women across OECD Countries." *Oxford Review of Economic Policy* 21 (3): 416–38.
- Fuchs-Schündeln, Nicola, and Matthias Schündeln. 2020. "The Long-Term Effects of Communism in Eastern Europe." *Journal of Economic Perspectives* 34 (2): 172–91.
- Fukase, Emiko. 2013. "Export Liberalization, Job Creation, and the Skill Premium: Evidence from the US-Vietnam Bilateral Trade Agreement (BTA)." World Development 41 (1): 317-337.
- Gaddis, Isis, and Stephan Klasen. 2014. "Economic Development, Structural Change, and Women's Labor Force Participation: A Reexamination of the Feminization U Hypothesis." *Journal of Population Economics* 27: 639-681.
- Gaddis, Isis, and Janneke Pieters. 2017. "The Gendered Labor Market Impacts of Trade Liberalization. Evidence from Brazil." *Journal of Human Resources* 52 (2): 457–490.

- Gendreau, Francis, Tien Dung Do, and Do Nhat Tan Pham. 2000. "Les Migrations Internes."
 In *Population et Développment Au Viêtnam*, edited by Patrick Gubry, 195–213.
 Clamecy: Karthala.
- General Statistics Office. 1991. "Vietnam Population Census 1989: Detailed Analysis of Sample Results." Hanoi, Vietnam.
- Giuliano, Paola. 2007. "Living Arrangements in Western Europe: Does Cultural Origin Matter?." *Journal of the European Economic Association* 5 (5): 927–52.
- Goldin, Claudia. 1995. "The U-Shaped Female Labor Force Function in Economic Development and Economic History." In Schultz TP Investment in Women's Human Capital and Economic Development, 61–90. University of Chicago Press.
- Goldsmith-Pinkham, Paul, Isaac Sorkin, and Henry Swift. "Bartik Instruments: What, When, Why, and How." *American Economic Review* 110, no. 8 (2020): 2586-2624.
- Goodkind, Daniel. 1995. "Rising Gender Inequality in Vietnam since Reunification." *Pacific Affairs* 68 (3): 342–57.
- Gourou, Pierre. 1936. "Les Paysans Du Delta Tonkinois: Étude de Géographie Humaine: Thèse Pour Le Doctorat Ès-Lettres Présentée à La Faculté Des Lettres de l'Université de Paris." *Journal of the Royal Asiatic Society*. Paris: Université de Paris.
- Haddad, Joanne, and Lamis Kattan. 2024. "Female-Specific Labor Regulation and Employment: Historical Evidence from the United States." *Journal of Labor Economics*, forthcoming.
- Hoang, Trung X., and Ha M. Nguyen. 2020. "Impact of US Market Access on Local Labour Markets in Vietnam." *Economics of Transition and Institutional Change* 28 (2): 315-343.
- Ichino, Andrea, and Giovanni Maggi. 2000. "Work Environment and Individual Background: Explaining Regional Shirking Differentials in a Large Italian Firm." *Quarterly Journal*

of Economics 115 (3): 1057-1090.

- International Labour Organization. "ILO Modelled Estimates and Projections database (ILOEST)". *ILOSTAT*. Accessed March 01, 2022. https://ilostat.ilo.org/data/.
- Jayachandran, Seema. 2015. "The Roots of Gender Inequality in Developing Countries." Annual Review of Economics 7 (1): 63–88.
- Jones, Gavin W. 1982. "Population Trends and Policies in Vietnam." *Population and Development Review* 8 (4): 783–810.
- Juhn, Chinhui, Gergely Ujhelyi, and Carolina Villegas-Sanchez. 2013. "Trade Liberalization and Gender Inequality." *American Economic Review*, 103 (3): 269–73.
- Keller, Wolfgang, and Hale Utar. 2022. "Globalization, Gender, and the Family." *Review of Economic Studies*, 89 (6): 3381–3409.
- Klasen, Stephan. 2019. "What Explains Uneven Female Labor Force Participation Levels and Trends in Developing Countries?." *The World Bank Research Observer* 34 (2): 161-197.
- Kovak, Brian K. 2013. "Regional Effects of Trade Reform: What Is the Correct Measure of Liberalization?." American Economic Review 103 (5): 1960–76.
- Kreibaum, Merle, and Stephan Klasen. 2015. "Missing Men: Differential Effects of War and Socialism on Female Labour Force Participation in Vietnam." Discussion Papers, No. 181. Göttingen.
- LaFont, Suzanne. 2001. "One Step Forward, Two Steps Back: Women in the Post-Communist States." *Communist and Post-Communist Studies* 34 (2): 203–20.
- Le, Hong Hiep. 2012. "Performance-Based Legitimacy: The Case of the Communist Party of Vietnam and 'Doi Moi." *Contemporary Southeast Asia* 34 (2): 145–72.
- Lippmann, Quentin, Alexandre Georgieff, and Claudia Senik. 2020. "Undoing Gender with Institutions: Lessons from the German Division and Reunification." *Economic Journal*

130 (629): 1445–1470.

- Ly, Hieu Tung. 2021. "Impacts of Confucianism on Vietnamese Culture." *International Communication of Chinese Culture* 8 (3).
- Mai, Thi Tu, and Thi Nam Tuyet Le. 1976. *La Femme Au Viemam*. Hanoi: Foreign Languages Publishing House.
- Mammen, Kristin, and Christina Paxson. 2000. "Women's Work and Economic Development." *Journal of Economic Perspectives* 14 (4): 141-164.
- Maurin, Eric, and Julie Moschion. 2009. "The Social Multiplier and Labor Market Participation of Mothers." *American Economic Journal: Applied Economics* 1 (1): 251-272.
- McCaig, Brian. 2011. "Exporting out of Poverty: Provincial Poverty in Vietnam and U.S. Market Access." *Journal of International Economics* 85 (1): 102-113.
- McCaig, Brian, and Nina Pavenik. 2015. "Informal Employment in a Growing and Globalizing Low-Income Country." *American Economic Review* 105 (5): 545-50.
- McCaig, Brian, and Nina Pavcnik. 2018. "Export Markets and Labor Allocation in a Low-Income Country." *American Economic Review* 108 (7): 1899-1941.
- McCaig, Brian, Nina Pavcnik, and Woan Foong Wong. 2023. "FDI Inflows and Domestic Firms: Adjustments to New Export Opportunities." NBER Working Paper 30729.
- Ngai, L. Rachel, and Barbara Petrongolo. 2017. "Gender Gaps and the Rise of the Service Economy." *American Economic Journal: Macroeconomics* 9 (4): 1-44.
- Ngai, L. Rachel, Claudia Olivetti and Barbara Petrongolo. 2024. "Gendered Change: 150 Years of Transformation in US Hours." NBER Working Paper 32475.
- Nguyen, Nhat Tuyen. 1999. "Transitional Economy, Technological Change and Women's Employment: The Case of Vietnam." *Gender, Technology and Development* 3 (1): 43–64.

Olivetti, Claudia. 2013. "The Female Labor Force and Long-Run Development: The American

Experience in Comparative Perspective." NBER Working Paper 19131.

- Olivetti, Claudia, Eleonora Patacchini, and Yves Zenou. 2020. "Mothers, Peers, and Gender-Role Identity." *Journal of the European Economic Association* 18 (1): 266–301.
- Tarp, Finn. 2018. "Viet Nam The Dragon That Rose from the Ashes." 126. WIDER Working Paper 2018. Helsinki.
- Topalova, Petia. 2010. "Factor Immobility and Regional Impacts of Trade Liberalization:
 Evidence on Poverty from India." *American Economic Journal: Applied Economics* 2 (4): 1–41.
- Tri, Vo Nhan. 1988. "Party Policies and Economic Performance: The Second and Third Five-Year Plans Examined." In *Postwar Vietnam: Dilemmas in Socialist Development*, edited by David G. Marr and Christine P. White, 77–90. Cornell University Press.
- Vu, Tien Manh, and Hiroyuki Yamada. 2023. "The impacts of Confucianism on gender inequality in Vietnam." *Journal of Economic Inequality*.
- Wang, Feicheng, Krisztina Kis-Katos, and Minghai Zhou. 2022. "Trade Liberalization and the Gender Employment Gap in China." *Journal of Human Resources* 58 (1).
- Werner, Jayne. 1981. "Women, Socialism, and the Economy of Wartime North Vietnam, 1960-1975." *Studies in Comparative Communism* 14 (2–3): 165–90.
- Werner, Jayne Susan, and Danièle Bélanger. 2002. Gender, Household, State. Đổi Mới in Việt
 Nam. Edited by Danièle Bélanger and Jayne Susan Werner. Southeast Asia Program,
 Cornell University.
- Whitmore, John K. 2023. "Confucianism in Vietnam." In *The Oxford Handbook of Confucianism*, edited by Jennifer Oldstone-Moore, online edn, 325–37.

		No	rth	So	uth
	(1) All	(2) Women	(3) Men	(4) Women	(5) Men
Panel A. All individuals aged 20-64					
Female	0.51	1.00	0.00	1.00	0.00
	(0.50)	(0.00)	(0.00)	(0.00)	(0.00)
Year of birth	1971.69	1971.21	1971.70	1971.40	1972.42
	(13.03)	(13.19)	(13.23)	(12.94)	(12.79)
Age	39.87	40.22	39.78	40.21	39.30
Ŭ	(12.31)	(12.37)	(12.40)	(12.26)	(12.19)
Married	0.84	0.89	0.83	0.85	0.78
	(0.37)	(0.31)	(0.37)	(0.36)	(0.42)
Ethnic Viet (majority)	0.83	0.74	0.73°	0.91	0.91
	(0.38)	(0.44)	(0.45)	(0.29)	(0.29)
Education:	· /	· /	· /	· · · ·	· · /
None	0.20	0.16	0.12	0.29	0.20
	(0.40)	(0.36)	(0.32)	(0.45)	(0.40)
Primary school	0.24	0.18	0.16	0.31	0.29
·	(0.43)	(0.38)	(0.37)	(0.46)	(0.46)
Lower secondary school	0.26	0.34	0.33^{-1}	0.17	0.21
v	(0.44)	(0.48)	(0.47)	(0.38)	(0.41)
Upper secondary/Vocational training or more	0.31	0.32	0.39	0.24	0.29
0, 0	(0.46)	(0.47)	(0.49)	(0.42)	(0.46)
Work in the last 12 months	0.89	0.91	0.92	0.82	0.92
	(0.31)	(0.28)	(0.27)	(0.38)	(0.28)
Observations	402833	94795	90544	110788	106706

Table 1: Characteristics of working age population in North and South Vietnam

Panel B. Conditional on having worked in the past 12 months

•					
Formal sector	0.22	0.21	0.23	0.22	0.24
	(0.42)	(0.40)	(0.42)	(0.41)	(0.42)
Working in foreign firms	0.03	0.04	0.02	0.04	0.02
	(0.17)	(0.19)	(0.13)	(0.20)	(0.15)
Industry:					
Agriculture	0.48	0.57	0.46	0.43	0.46
	(0.50)	(0.49)	(0.50)	(0.50)	(0.50)
Manufacturing	0.14	0.14	0.14	0.16	0.12
	(0.35)	(0.35)	(0.34)	(0.37)	(0.33)
Construction	0.07	0.02	0.15	0.01	0.11
	(0.26)	(0.15)	(0.35)	(0.11)	(0.31)
Services	0.31	0.26	0.26	0.39	0.31
	(0.46)	(0.44)	(0.44)	(0.49)	(0.46)
Occupation:					
Elementary occupations	0.54	0.64	0.53	0.53	0.49
	(0.50)	(0.48)	(0.50)	(0.50)	(0.50)
Medium skilled occupations	0.36	0.26	0.38	0.38	0.42
	(0.48)	(0.44)	(0.49)	(0.49)	(0.49)
Managers/Professionals	0.09	0.10	0.09	0.09	0.09
	(0.29)	(0.29)	(0.28)	(0.29)	(0.29)
Observations	358806	86520	83246	91290	97750

Notes: Mean characteristics of individuals aged 20-64 in VHLSS 2002-2018 are reported. Panel B includes those who worked in the last 12 months. Armed forces occupations, representing 0.3% of total workforce, are excluded from the analysis. Standard deviations in parentheses.

	Γ	Dependent v	variable: W	fork
	(1)	(2)	(3)	(4)
Exposure	-0.007 (0.005)	-0.013^{***} (0.004)	$0.003 \\ (0.005)$	
Female \times Exposure			-0.031^{***} (0.007)	-0.031^{***} (0.007)
Province FE	Yes	Yes	Yes	No
Year FE	Yes	Yes	No	No
Gender-by-year FE	No	No	Yes	Yes
Province-by-year FE	No	No	No	Yes
Province initial characteristics \times year FE				
Share of manufacturing employment	Yes	Yes	Yes	No
Manufacturing gross output per capita	No	Yes	Yes	No
Service gross output per capita	No	Yes	Yes	No
Female labor force participation	No	Yes	Yes	No
Kindergarten classes per 0-5 child	No	Yes	Yes	No
Mean dep. var.	0.891	0.891		
Mean dep. var Men			0.918	0.918
Mean dep. var Women			0.865	0.865
Observations	402,833	402,833	402,833	402,833
R^2	0.128	0.129	0.131	0.132

Table 2: Provincial exports and employment responses

Notes: "Work" is a dummy indicating whether the individual worked at any time during the past 12 months. The sample includes all individuals aged 20-64. All columns control for age in cubic polynomial, gender, education level, ethnicity, province fixed effects, year fixed effects, and interactions of provinces' initial characteristics with year dummies. Provinces' initial characteristics are taken from the Vietnam Statistical Yearbook 1999. Columns 3 and 4 allow for gender-by-year fixed effects. Column 4 allows for province-by-year fixed effects. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

				Dependent variable: Work	variable: V	Vork		
	1ra 2002	Traded 2002-2018	Manufa 2002	Manufacturing 2002-2018	$^{\mathrm{Tr}_{6}}_{2010}$	Traded 2010-2018	Log Gl 201(Log GRDP p/c 2010-2018
	(1)	(2)	(3)	(4)	(5)	(9)	(2)	(8)
Exposure	-0.013^{***} (0.004)	0.003 (0.005)	-0.007 (00.09)	0.007 (0.008)	-0.009 (0.008)	0.006 (0.07)		
Female \times Exposure		-0.031^{***} (0.007)		-0.027^{***} (0.005)		-0.029^{***} (0.007)		
Log GRDP per capita							-0.024^{*} (0.014)	-0.009 (0.015)
Female \times Log GRDP per capita								-0.030^{*} (0.016)
Mean dep. var Men	0.918	0.918	0.918	0.918	0.917	0.917	0.917	0.917
Mean dep. var Women Observations	$0.865 \\ 402,833$	$0.865 \\ 402,833$	$0.865 \\ 402,833$	$0.865 \\ 402,833$	0.859 270,731	0.859 270,731	0.859 270,731	0.859 270,731
R^{2}	0.129	0.131	0.129	0.131	0.131	0.132	0.131	0.131
Notes: Columns 1 and 2 report the baseline results from columns 2 and 3 of Table 2, where Exposure is export exposure driven by all traded industries. In columns 3 and 4, Exposure is export exposure driven by manufacturing industries. In columns 5 to 8, the time frame is restricted to 2010-2018. Columns 5 and 6 report the effects of export exposure driven by all traded industries. Columns 7 and 8 report the effects of log GRDP per capita. "Work" is a dummy indicating whether the individual worked at any time during the past 12 months. The sample includes all individuals aged 20-64. All columns control for age in cubic polynomial, gender, education level, ethnicity, province fixed effects, year fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Columns 2, 4, 6, and 8 allow for gender-by-year fixed effects. Standard errors are clustered at the province level, * $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$.	eline results f e is export ex the effects of indicating wh control for ag itial characte rors are clust	rom columns posure driver export expose tether the in- in cubic po ristics with y ered at the p	2 and 3 of 1 by manufa ure driven b dividual wor olynomial, g rear dummic rovince level	Table 2, wh cturing indu- y all traded is ked at any ti ender, educa s (as in colu , * $p < 0.10$,	tere Exposunt stries. In co- industries. ime during it tion level, e $mn \ 3 \ of Ta$ *** $p < 0.05$	e is export e lumns 5 to 8 Columns 7 a the past 12 n thnicity, pro ble 2). Colus , *** $p < 0.0$	xposure drive, the time fra-nd 8 report tnonths. Thevince fixed $efmns 2, 4, 6,1.$	m by all traded me is restricted he effects of log sample includes fects, year fixed and 8 allow for

on employment
development on e
ndicators of de-
alternative indicat
Effect of a
Table 3:

	(3) ng Construction		Turt	Empioyer
-0.044^{**} (0.020) 0.479 358,806 0.285 0.285 0.285 0.285 0.285 0.285 0.285 0.285 0.285 (0.018) (0.010)		(4) Services	(5) Formal firm	(6) Foreign firm
-0.044^{**} (0.020) 0.479 358,806 0.285 0.285 0.285 0.285 0.018) -0.025^{**} (0.010)				
$\begin{array}{c} 0.479\\ 358,806\\ 0.285\\ 0.285\\ -0.032^{*}\\ (0.018)\\ -0.025^{**}\\ (0.010)\end{array}$	-0.002 (0.004)	0.026^{***} (0.007)	0.024^{*} (0.013)	0.019^{*} (0.010)
358,806 0.285 -0.032* (0.018) -0.025** (0.010)	0.072	0.307	0.223	0.031
0.285 -0.032* (0.018) -0.025** (0.010)	358,806	358,806	358, 338	358, 338
-0.032^{*} (0.018) -0.025^{**} (0.010)	0.072	0.185	0.285	0.093
(0.018) -0.025** (0.010)	0.003	0.016^{**}	0.024^{*}	0.012
-0.025^{**} (0.010)	(0.005)	(0.008)	(0.012)	(0.010)
	-0.009 (0.006)	0.019^{***} (0.006)	0.000 (0.006)	0.014^{***} (0.005)
0.458	0.127	0.286	0.232	0.021
Observations - Vomen 0.501 0.154 Observations 358,806 358,806	0.017 358,806	0.328 358,806	0.213 $358,338$	0.041 $358,338$
R^2 0.285 0.090	0.073	0.186	0.285	0.096

	Dependent	variable: Work
	(1)	(2)
Single		
Exposure	0.016^{*} (0.009)	0.018^{**} (0.009)
Female \times Exposure		-0.006 (0.006)
Married		
Exposure	-0.019^{***} (0.004)	-0.002 (0.005)
Female \times Exposure		-0.033*** (0.008)
Mean dep. var.	0.891	0.891
Observations R^2	$\begin{array}{c} 402,833 \\ 0.143 \end{array}$	$402,833 \\ 0.145$

Table 5: Heterogeneous effects by marital status

Notes: "Work" is a dummy indicating whether the individual worked at any time during the past 12 months. The sample includes all individuals aged 20-64. All columns control for age in cubic polynomial, gender, education level, ethnicity, province fixed effects, marital-status-by-year fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Column 2 allows for gender-by-year fixed effects and for gender-by-marital-status fixed effects. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

		Depe	Dependent variable: Work	ork	
	(1) Household wealth	(2) Husband hourly wage	(3) Husband annual earnings	(4) Husband education	(5) Husband occupation
Exposure	-0.006 (0.006)	-0.005 (0.007)	-0.005 (0.007)	-0.014^{***} (0.005)	-0.007 (0.005)
$Q2 \times Exposure$	-0.000 (0.003)	-0.000 (0.006)	0.001 (0.005)		
$Q3 \times Exposure$	-0.007 (0.004)	-0.001 (0.005)	-0.001 (0.005)		
$Q4 \times Exposure$	-0.013^{**} (0.006)	-0.015^{***} (0.004)	-0.014^{**} (0.005)		
Medium × Exposure				0.005 (0.003)	-0.012^{***} (0.003)
High \times Exposure				-0.004 (0.007)	-0.013 (0.010)
Mean dep. var. Observations R^2	$\begin{array}{c} 0.891 \\ 177,778 \\ 0.132 \end{array}$	$\begin{array}{c} 0.909 \\ 45,171 \\ 0.118 \end{array}$	$\begin{array}{c} 0.909 \\ 45,171 \\ 0.120 \end{array}$	$\begin{array}{c} 0.927 \\ 118,673 \\ 0.088 \end{array}$	0.927 118,673 0.090
Notes: The sample is restricted to married women aged $20-64$. Column 1 include married women with information on hourly household wealth, columns 2 and 3 include married women whose husbands do waged work with information on hourly wage and annual earnings, columns 4 and 5 include married women with information on husbands' education and occupation skill level. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Additionally, columns 1 to 3 control for family SES quartile-specific year fixed effects (type of family SES is indicated at the top of the column), the reference group is Q1. Columns 4 controls husband education level-specific year fixed effects, the reference group is low education. Standard errors are clustered at province level, * $p < 0.10$, *** $p < 0.05$, *** $p < 0.01$.	to married women 1 3 include marrie imns 4 and 5 inc is control for age nitial characterist ly SES quartile-s oup is Q1. Colum Columns 5 contr andard errors are	a aged 20-64. Co. d women whose i lude married wo in cubic polynom ics with year du pecific year fixed nns 4 controls hu ol for husband oc clustered at prov	with 1 include marri- iusbands do waged we men with informatio, ial, education level, e mmies (as in column effects (type of fami isband education level upation skill-specific ince level, * $p < 0.10$,	ad women with wrk with inform n on husbands thnicity, provi β of Table 2, ly SES is indi β -specific year J year fixed effec ** $p < 0.05$,	information on ation on hourly ' education and nce fixed effects, I. Additionally, cated at the top fixed effects, the ts, the reference ** p < 0.01.

naracteristics
5
husband's
and
by wealth
by
effects
Heterogeneous ef
6:
Table

			Dependent variable: Work	riable: Wo	rk	
		Whole sample	ile		Married	
	(1) All	(2) Excl. 12 provinces in Mekono River Delta	(3) Excl. 15 provinces on Khmer side me-1833	(4) A11	(5) Excl. 12 provinces in Mekonø River Delta	(6) Excl. 15 provinces on Khmer side nre-1833
		BATO TO INT QUINTIN			MATO TO TATA QUINTING	and and and
North Exposure (α_0)	-0.005 (0.005)	-0.004 (0.005)	-0.005 (0.005)	-0.005 (0.004)	-0.004 (0.005)	-0.004 (0.005)
Female × Exposure (α_1)	-0.013^{***} (0.002)	-0.015^{***} (0.002)	-0.015^{***} (0.002)	-0.017^{***} (0.003)	-0.018^{***} (0.002)	(0.002)
South	~		~	~	~	
Exposure (β_0)	-0.008 (0.006)	-0.005 (0.005)	-0.003 (0.005)	0.003 (0.006)	0.005 (0.005)	0.006 (0.006)
Female × Exposure (β_1)	-0.022^{***} (0.006)	-0.031^{***} (0.005)	-0.035^{***} (0.005)	-0.029^{***} (0.007)	-0.038^{***} (0.006)	-0.042^{***} (0.006)
$\beta_0 - \alpha_0$	-0.003	-0.001	0.001	0.008	0.009	0.010*
$eta_1 - lpha_1$	(0.005)	(0.005) -0.017 + + (0.005)	(0.004) -0.020*** (0.004)	(0.000) -0.012* (0.006)	(0.002) - 0.020 ***	(0.000) -0.023*** (0.005)
Mean dep. var. Observations	$0.891 \\ 402,833$	0.897 317,355	0.898 299,693	0.920 336,504	0.930 265,385	0.931 250,942
R^2	0.134	0.145	0.148	0.124	0.128	0.129
Notes: 12 provinces in Mekong River Delta include Long An, Tien Giang, Ben Tre, Tra Vinh, Vinh Long, Dong Thap, An Giang, Kien Giang, Can Tho, Hau Giang, Soc Trang, Bac Lieu, Ca Mau. 15 provinces on the Khmer side pre-1833 include the 12 provinces in Mekong River Delta and Tay Ninh, Binh Duong, Binh Phuoc. The sample is restricted to working individuals aged 20-64 in columns 1 to 3, and married working individuals aged 20-64 in columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, gender-by-region fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Standard errors are clustered at the province level, * $p < 0.10$, *** $p < 0.05$, *** $p < 0.01$.	River Delta in 15 provinces ing individuals in level, ethnic s (as in colum	<pre>slude Long An, Tien Giang, on the Khmer side pre-183: aged 20-64 in columns 1 t ity, province fixed effects, ge n 3 of Table 2). Standard en</pre>	Tien Giang, Ben Tre, Tra Vinh, Vinh Long, Dong Thap, An Giang, Kien Giang, Can Tho, Hau Giang, side pre-1833 include the 12 provinces in Mekong River Delta and Tay Ninh, Binh Duong, Binh Phuoc. c columns 1 to 3, and married working individuals aged 20-64 in columns 4 to 6. All columns control for we effects, gender-by-year fixed effects, gender-by-region fixed effects, and interactions of provinces' initial d. Standard errors are clustered at the province level, * $p < 0.10$, *** $p < 0.05$, *** $p < 0.01$.	ong, Dong '. 1 Mekong Ri 1.dividuals ag 1.ader-by-regi 1.vince level, *	Thap, An Giang, Kien Gian ver Delta and Tay Ninh, B ed 20-64 in columns 4 to 6. on fixed effects, and interac p < 0.10, ** p < 0.05, *** 1	g, Can Tho, Hau Giang, inh Duong, Binh Phuoc. All columns control for tions of provinces' initial > < 0.01.

Table 7: Employment responses across various choices of territory

			Dependent variable: Work	ariable: Wo	ırk	
		Traded			Manufacturing	60
	(1) All	(2) Female-intensive	(3) Male-intensive	(4) All	(5) Female-intensive	(6) Male-intensive
North						
Exposure (α_0)	-0.005	-0.007	0.004	-0.007	-0.007	0.007
	(0.004)	(0.006)	(0.003)	(0.006)	(0.006)	(0.004)
Female × Exposure (α_1)	-0.017***	-0.018***	-0.011***	-0.012^{***}	-0.011***	-0.010^{***}
	(0.003)	(0.003)	(0.003)	(200.0)	(0.002)	(0.003)
\mathbf{South}						
Exposure (β_0)	0.003	-0.008	0.009	0.002	-0.006	0.011
	(0.006)	(0.011)	(0.006)	(0.010)	(0.010)	(0.008)
Female \times Exposure (β_1)	-0.029^{***}	-0.026^{***}	-0.024^{***}	-0.026^{***}	-0.024^{***}	-0.019^{*}
	(0.007)	(0.008)	(0.007)	(0.007)	(0.007)	(0.010)
$\beta_0 - \alpha_0$	0.008	-0.001	0.004	0.009	0.001	0.004
	(0.006)	(0.007)	(0.006)	(0.006)	(0.006)	(0.007)
eta_1-lpha_1	-0.012^{*}	-0.00	-0.013^{**}	-0.014^{**}	-0.012^{**}	-0.009
	(0.006)	(0.007)	(0.007)	(0.006)	(0.006)	(0.009)
Mean dep. var.	0.920	0.920	0.920	0.920	0.920	0.920
Observations	336,504	336,504	336,504	336,504	336,504	336,504
R^2	0.124	0.124	0.124	0.124	0.124	0.123
Notes: Column 1 reports the baseline results from column 4 of Table 7. In columns 1 to 3, Exposure is export exposure driven by traded industries. In columns 4 to 6, Exposure is export exposure driven by manufacturing industries. Columns 1 and 4 use export exposure driven by all industries, columns 2 and 5 use female-intensive industries, and columns 3 and 6 use male-intensive industries. An industry is female-intensive if the share of female workers in that industry is above median (36%) in 1999, and male-intensive otherwise. The sample is restricted to married individuals aged 20-64. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, gender-by-year fixed effects, gender- by-region fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Columns 2, 3, 5, and 6 additionally control for interactions of provinces' initial share of employment in female-intensive manufacturing with year dummies. Standard	line results fron port exposure a sive industries y is above med or age in cubic ctions of provin	e results from column 4 of Table 7. In colum t exposure driven by manufacturing indust se industries, and columns 3 and 6 use male s above median (36%) in 1999, and male-in age in cubic polynomial, education level, et ons of provinces' initial characteristics with a of provinces' initial share of employmem	In columns 1 to 3, 1 9 industries. Colum use male-intensive oth 1 male-intensive oth 1 level, ethnicity, pro itics with year dumm ployment in female-	Exposure is e. ns 1 and 4 us industries. A erwise. The . vince fixed e! nies (as in co intensive ma	port exposure driven l ie export exposure driv n industry is female-in sample is restricted to fects, gender-by-year f lumn 3 of Table 2). C nufacturing with year	y traded industries en by all industries utensive if the shar married individual ixed effects, gender olumns 2, 3, 5, and dummies. Standar

Table 8: Employment responses driven by female- vs. male-intensive industry

		Dependent v	variable: Wo	ork
	All Southe	ern provinces	Destinati	on provinces
	(1) All	(2) Migrants	(3) All	(4) Migrants
Southern-born				
Exposure (α_0)	-0.004 (0.020)	-0.085 (0.055)	-0.193 (0.145)	-0.192 (0.237)
Female × Exposure (α_1)	-0.038^{***} (0.007)	-0.035^{***} (0.007)	-0.036^{***} (0.007)	-0.030^{**} (0.009)
Northern-born				
Exposure (β_0)	-0.002 (0.021)	-0.088 (0.055)	-0.194 (0.145)	-0.194 (0.236)
Female × Exposure (β_1)	-0.024^{***} (0.005)	-0.024^{***} (0.005)	-0.021^{***} (0.006)	-0.021^{***} (0.006)
$\beta_0 - \alpha_0$	$0.002 \\ (0.004)$	-0.003 (0.003)	-0.001 (0.004)	-0.002 (0.003)
$\beta_1 - \alpha_1$	0.014^{*} (0.007)	0.010*** (0.004)	0.014^{**} (0.006)	0.009^{*} (0.004)
Mean dep. var.	0.880	0.877	0.900	0.905
Observations R^2	$72,901 \\ 0.114$	$17,\!430 \\ 0.147$	$17,934 \\ 0.109$	$11,\!205 \\ 0.115$

Table 9: Employment responses by birth origin within the South, cohorts 1954-1979

Notes: In columns 1 and 2, the sample includes all 32 provinces in the South. In columns 3 and 4, the sample includes eight "Destination" provinces in the South-those with highest shares of Northernborn immigrants. The sample is restricted to individuals aged 20-64 from 1954-1979 birth cohorts. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, gender-by-year fixed effects, gender-by-birth-origin fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Columns 1 and 2 control for for gender-by-birth-origin-by-destination fixed effects. Birth origin is an indicator for being born in the North. Destination is an indicator for living in a Destination province. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.



Figure 1: Female to male labor force participation rate across countries

Notes: Southeast Asia includes Brunei, Cambodia, Indonesia, Laos, Malaysia, Myanmar, Philippines, Singapore, Thailand, Timor-Leste, and Vietnam. Sources: International Labor Organization.



Figure 2: Labor force participation rate by gender and region in Vietnam, 2002-2018

Notes: This figure plots the share of individuals aged 20-64 who reported to have worked in the last 12 months. Source: VHLSS 2002-2018.





Notes: N = 120. Observations at province-year level, data in 2010-2018. Data on provinces' Gross Regional Domestic Product per capita (GRDP per capita) are from the General Statistics Office of Vietnam (GSO). Export exposure is log predicted export per worker, as defined in equation (1). The plots partial out year-specific effects of a province's share of manufacturing employment in 1999. Regression models are weighted by provinces' share of national population in 1999. A province's share of manufacturing employment in 1999 and share of the national population in 1999 are taken from the Census 1999.

Figure 4: Spatial variation in export exposure between 2002 and 2018



(a) Export exposure in 2002 and 2018



(b) Change in export exposure, 2002-2018

Notes: Export exposure is log predicted export per worker, as defined in equation (1). Export values (constant 2018 VND) from United Nations Comtrade Database (UN Comtrade) are adjusted for the Consumer Price Index obtained from the General Statistics Office (GSO) of Vietnam. In Panel B, the solid purple line represents the 17th parallel north, the Demilitarized Zone dividing North and South Vietnam.

Figure 5: Opinions of Southern Vietnamese compared to Northern Vietnamese



Notes: Coefficients come from separate regressions of each outcome (opinion) on the South indicator, controlling for year fixed-effects, with confidence intervals at 95% level. Answers are recoded from "Strongly Agree", "Agree" and "Neither" into 1 (Agree); "Disagree" and "Strongly Disagree" into 0 (Disagree). Source: World Values Survey 2001, 2006 and 2019.



Figure 6: Heterogeneous effects by quartile of wealth and husband earnings



Notes: The sample is restricted to married women aged 20-64. Coefficients (with 95% confidence intervals) are obtained from a regression where we interact $Exposure_{pt}$ in equation (2) with quartile of wealth or husband earnings and indicator for South, controlling for age in cubic polynomial, education level, ethnicity, province fixed effects, interactions of provinces' initial characteristics with year fixed effects, quartile-by-region fixed effects, and quartile-specific year fixed effects. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.



Figure 7: Heterogeneous effects by husband education and occupation skill

Notes: The sample is restricted to married women aged 20-64. Coefficients (with 95% confidence intervals) are obtained from a regression where we interact $Exposure_{pt}$ in equation (2) with level of husband education or husband occupation skill and indicator for South, controlling for age in cubic polynomial, education level, ethnicity, province fixed effects, interactions of provinces' initial characteristics with year fixed effects, husband-education/occupation-by-region fixed effects, and husband-education/occupation year fixed effects. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

Figure 8: Northern-born and immigrant shares by provinces in South Vietnam



Notes: Northern born are people whose birth province belongs to former North Vietnam. Immigrants are those born in a province different from the current province, regardless of whether the birth province is in North or South Vietnam. Source: VHLSS 2014 – 2018.

Appendix A

Table A1: Summary statistics of Exposure and growth of exports across provinces between 2002 and 2018

Panel A. Descriptive statistics	s of Exposure	е			
	Mean	Std. dev.	Min	Max	Obs.
Exposure	17.271	0.952	15.046	20.413	540

Panel B. 7	Fop 10 largest export	ting province	es by year			
	Year 2002			Year 2018		
Rank	Province	Exposure	Export per worker (million VND)	Province	Exposure	Export per worker (million VND)
1	Ba Ria - Vung Tau	19.721	367.109	Ho Chi Minh City	20.353	690.453
2	Ho Chi Minh City	18.252	84.497	Ha Noi, Ha Tay	19.978	474.519
3	Dong Nai	17.833	55.582	Dong Nai	19.927	451.059
4	Binh Duong	17.731	50.173	Binh Duong	19.864	423.333
5	Da Nang	17.537	41.318	Thai Nguyen	19.433	275.26
6	Hai Phong	17.237	30.61	Ba Ria - Vung Tau	19.351	253.601
7	Ha Noi, Ha Tay	17.199	29.486	Hai Phong	19.275	235.005
8	Khanh Hoa	17.057	25.563	Da Nang	19.214	220.963
9	Quang Ninh	17.028	24.842	Quang Ninh	18.997	178.002
10	Thua Thien-Hue	16.992	23.965	Long An	18.692	131.162

Panel C. Top 10 largest change in Exposure in period 2002-2018

	op to migore enume	, <u> </u>	$\Delta \tilde{E}$ xport
			per worker
Rank			(million
$\Delta Exposure$	Province	$\Delta Exposure$	VND)
1	Thai Nguyen	3.398	266.056
2	Ha Noi, Ha Tay	2.778	445.033
3	Phu Tho	2.349	78.256
4	Bac Giang	2.231	43.885
5	Hung Yen	2.138	68.03
6	Binh Duong	2.133	373.16
7	Ben Tre	2.13	81.395
8	Ho Chi Minh City	2.101	605.957
9	Dong Nai	2.094	395.477
10	Hai Phong	2.038	204.395

Notes: Observations at province-year level, with 60 provinces in 9 time periods. Exposure is constructed according to equation (1). Export values (constant 2018 VND) from United Nations Comtrade Database (UN Comtrade) are adjusted for Consumer Price Index obtained from the General Statistics Office (GSO) of Vietnam.

Table A2: Population by regions before and after the 1954 division

	$\begin{array}{c} {\rm Tonkin} \\ ({\rm north}) \end{array}$	Annam (center)	$\begin{array}{c} { m Cochinchina} \\ { m (south)} \end{array}$	Vietnam (total)
Gross birth rate	37.8	29.6	37	37.5
Gross death rate	22-24	17.6	24.1	24.2
Growth rate	14	12	12.9	13.3

Panel A. Population growth rate before 1945

Panel B. Population before the 1954 division (in millions)

	Tonkin (north)	$\begin{array}{c} \text{Annam} \\ (\text{center}) \end{array}$	$\begin{array}{c} { m Cochinchina} \\ { m (south)} \end{array}$	Vietnam (total)
1901	5.5	4.5	3	13
1911	6.1	5.5	3.8	14.7
1921	6.9	4.9	3.8	15.8
1926	7.4	5.6	4.1	17.1
1931	8.1	5.1	4.8	17.7
1936	8.7	5.7	4.6	19
1943	9.8	7.2	5.6	22.6

Panel C. Population after the 1954 division (in millions)

	North Vietnam	South Vietnam	Vietnam
1955	13.6	13.6	27.2

Sources: Banens (2000) (constructed from Indochina Statistical Yearbooks and Vietnam Statistical Yearbooks).

	Depe	ndent vari	able: Log a	nnual earnings
	Mair	n job	All jobs	s (incl. benefits)
	(1)	(2)	(3)	(4)
Exposure	$\begin{array}{c} 0.088^{***} \\ (0.024) \end{array}$	$\begin{array}{c} 0.096^{***} \\ (0.025) \end{array}$	$\begin{array}{c} 0.104^{***} \\ (0.034) \end{array}$	$\begin{array}{c} 0.103^{***} \\ (0.035) \end{array}$
Female \times Exposure		-0.018 (0.012)		$0.003 \\ (0.014)$
Mean dep. var Men Mean dep. var Women Observations R^2	$10.546 \\ 10.395 \\ 137,802 \\ 0.461$	$10.546 \\ 10.395 \\ 137,802 \\ 0.461$	$10.320 \\ 10.173 \\ 168,178 \\ 0.455$	$10.320 \\ 10.173 \\ 168,178 \\ 0.455$

Table A3: Wage effect by gender

Notes: The dependent variable in columns 1 and 2 is log annual salary from the main job, in columns 3 and 4 is log annual earnings from all jobs including bonuses and benefits. The sample includes individuals aged 20-64 working for wage or salary. All columns control for age in cubic polynomial, gender, education level, ethnicity, province fixed effects, year fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Columns 2 and 4 allow for gender-by-year fixed effects. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

			Dependent variable:	Log annu	Log annual earnings	
		Main job			All jobs (incl. benefits)	mefits)
	(1) All	(2) Excl. 12 provinces in Mekong River Delta	(3) Excl. 15 provinces on Khmer side pre-1833	(4) All	(5) Excl. 12 provinces in Mekong River Delta	(6) Excl. 15 provinces on Khmer side pre-1833
North						
Male × Exposure (α_0)	0.098^{***} (0.027)	0.109^{***} (0.033)	0.101^{***} (0.035)	0.111^{***} (0.036)	0.133^{***} (0.040)	0.110^{***} (0.034)
Female × Exposure (α_1)	0.063^{**} (0.027)	0.071^{**} (0.034)	0.066^{*} (0.036)	0.093^{***} (0.031)	0.115^{***} (0.034)	0.096^{***} (0.032)
South						
Male \times Exposure (β_0)	0.111^{***} (0.040)	0.129^{**} (0.048)	0.108^{**} (0.052)	0.098^{**} (0.046)	0.127^{**} (0.048)	0.090^{*} (0.049)
Female × Exposure (β_1)	0.127^{***} (0.038)	0.135^{***} (0.047)	0.114^{**} (0.049)	0.129^{***} (0.046)	0.156^{***} (0.046)	0.119^{**} (0.045)
$\beta_0 - \alpha_0$	0.013	0.021	0.008	-0.012	-0.006	-0.020
eta_1-lpha_1	(0.037) 0.064^{*}	(0.047) 0.064	(0.049) 0.048	(0.032) 0.036	(0.033) 0.040	(0.032) 0.023
4	(0.036)	(0.045)	(0.047)	(0.036)	(0.033)	(0.032)
Mean dep. var.	10.484	10.550	10.546	10.261	10.312	10.295
Observations	137,802	109,807	101,750	168, 178	134,427	125,546
R^{2}	0.462	0.457	0.459	0.455	0.474	0.475
Notes: 12 provinces in Mekong River Delta include Long An, Tien Giang, Ben Tre, Tra Vinh, Vinh Long, Dong Thap, An Giang, Kien Giang, Can Tho, Hau Giang, Soc Trang, Bac Lieu, Ca Mau. 15 provinces on the Khmer side pre-1833 include the 12 provinces in Mekong River Delta and Tay Ninh, Binh Duong, Binh Phuoc. Soc Trang, Bac Lieu, Ca Mau. 15 provinces on the Khmer side pre-1833 include the 12 provinces in Mekong River Delta and Tay Ninh, Binh Duong, Binh Phuoc. The dependent variable in columns 1 to 3 is log annual salary from the main job, in columns 4 to 6 is log annual earnings from all jobs including bonuses and benefits. The sample includes individuals aged 20-64 working for wage or salary. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, gender-by-region fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Standard errors are clustered at the province level, * $p < 0.10$, ** $p < 0.05$, *** $p < 0.01$.	River Delta in 15 provinces ns 1 to 3 is lo ns 1 to 3 is lo aged $20-64$ wo der-by-region f drec level, * p <	clude Long An, Tien Giang on the Khmer side pre-183 g annual salary from the m rking for wage or salary. A ixed effects, and interaction: 0.10, ** p < 0.05, *** p < 0	, Ben Tre, Tra Vinh, Vinh 3 include the 12 provinces ain job, in columns 4 to 6 i Il columns control for age in s of provinces' initial charac 0.01.	Long, Dong in Mekong I is log annual cubic polym teristics with	Thap, An Giang, Kien Gia River Delta and Tay Ninh, earnings from all jobs inclu omial, education level, ethni i year dummies (as in colum	ing, Can Tho, Hau Giang, Binh Duong, Binh Phuoc. Iding bonuses and benefits. Icity, province fixed effects, in 3 of Table 2). Standard

Table A4: Wage effect by gender and region

	Household wealth	Husband hourly wage	Husband annual earnings
	(1)	(2)	(3)
North			
Exposure (α_0)	$0.003 \\ (0.006)$	$0.004 \\ (0.008)$	$0.000 \\ (0.008)$
Q2 × Exposure (α_1)	0.001 (0.004)	-0.005 (0.006)	$0.001 \\ (0.006)$
Q3 × Exposure (α_2)	-0.010^{*} (0.006)	-0.008 (0.006)	$0.001 \\ (0.004)$
Q4 × Exposure (α_3)	-0.027^{***} (0.007)	-0.027^{***} (0.005)	-0.024^{***} (0.005)
South			
Exposure (β_0)	-0.009 (0.009)	-0.020^{**} (0.010)	-0.016^{*} (0.009)
Q2 × Exposure (β_1)	-0.003 (0.005)	$0.007 \\ (0.010)$	$0.003 \\ (0.006)$
Q3 × Exposure (β_2)	-0.006 (0.006)	$0.007 \\ (0.008)$	$0.004 \\ (0.008)$
Q4 × Exposure (β_3)	-0.007 (0.005)	-0.006 (0.007)	-0.004 (0.009)
$\beta_0 - \alpha_0$	-0.012* (0.007)	-0.024^{**} (0.009)	-0.016^{**} (0.007)
Mean dep. var. Observations R^2	$0.891 \\ 177,778 \\ 0.133$	$0.909 \\ 45,171 \\ 0.118$	$0.909 \\ 45,171 \\ 0.121$

Table A5: Heterogeneous effects by wealth and husband earnings: North-South difference

Notes: The sample is restricted to married women aged 20-64. Column 1 include married women with information on household wealth, columns 2 and 3 include married women whose husbands do waged work with information on hourly wage and annual earnings. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, interactions of provinces' initial characteristics with time dummies (as in column 3 of Table 2). Additionally, each column controls for family SES quartile × south fixed effects and quartile-specific year fixed effects (type of family SES is indicated at the top of the column). In each region, the reference group is Q1. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

	Depend	dent variable: Work
	Husband education	Husband occupation
	(1)	(2)
North		
Exposure (α_0)	-0.002 (0.006)	$0.002 \\ (0.006)$
Medium × Exposure (α_2)	-0.004 (0.004)	-0.019^{***} (0.003)
High \times Exposure (α_3)	-0.021^{***} (0.007)	-0.034^{***} (0.007)
South		
Exposure (β_0)	-0.016^{**} (0.008)	-0.012 (0.008)
Medium × Exposure (β_2)	$0.005 \\ (0.004)$	-0.007^{*} (0.004)
High \times Exposure (β_3)	$0.001 \\ (0.004)$	-0.006 (0.006)
$\beta_0 - \alpha_0$	-0.014** (0.006)	-0.014** (0.006)
Mean dep. var. Observations R^2	0.927 118,673 0.088	$0.927 \\ 118,673 \\ 0.091$

Table A6: Heterogeneous effects by husband education and occupation skill: North-South difference

Notes: The sample is restricted to married women aged 20-64 with information on husbands' education and occupation skill level. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Additionally, column 1 controls for husband education level \times south fixed effects and husband education level-specific year fixed effects, and column 2 controls for husband occupation skill level \times south fixed effects and husband occupation skill level \times south fixed effects. In each region, the reference group is low education (column 1) or low-skill occupation (column 2). Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.

Period	Objective	Achievement	Annual average
1976-1980	4,000,000	1,520,600	304,120
1981-1985	1,000,000	$1,\!257,\!300$	251,460
1986-1990	1,600,000	1,142,600	228,520
1991-1995	1,000,000	902,000	180,400
1996-2000	1,000,000	210,700*	$105,\!350^*$
Total	8,600,000	5,033,200**	239,700**

Table A7: Number of immigrants to New Economic Zones

Sources: Gendreau et al. (2000), compiling from Directorate of Migration and Construction of New Economic Zones, * data for 1996-97, ** data for 1976-1997.

Table A8: Employmen	responses by	birth origin	within the South,	cohorts 1954-1989
---------------------	--------------	--------------	-------------------	-------------------

	Dependent variable: Work				
	All Southe	ern provinces	Destination province		
	(1) All	(2) Migrants	(3) All	(4) Migrants	
Southern-born					
Exposure (α_0)	-0.014 (0.019)	-0.062 (0.045)	-0.101 (0.078)	-0.123 (0.156)	
Female × Exposure (α_1)	-0.031^{***} (0.006)	-0.028^{***} (0.006)	-0.030^{***} (0.007)	-0.027^{***} (0.007)	
Northern-born					
Exposure (β_0)	-0.012 (0.019)	-0.064 (0.045)	-0.103 (0.077)	-0.125 (0.156)	
Female × Exposure (β_1)	-0.020^{***} (0.005)	-0.020^{***} (0.005)	-0.020^{***} (0.006)	-0.020** (0.006)	
$\beta_0 - \alpha_0$	0.002	-0.002	-0.002	-0.002	
$\beta_1 - \alpha_1$	$(0.003) \\ 0.011^* \\ (0.007)$	(0.002) 0.008^{**} (0.003)	$(0.003) \\ 0.011^* \\ (0.005)$	$(0.003) \\ 0.007^* \\ (0.003)$	
Mean dep. var.	0.893	0.892	0.915	0.918	
Observations	100,873	22,636	25,727	$14,\!468$	
R^2	0.100	0.138	0.099	0.110	

Notes: In columns 1 and 2, the sample includes all 32 provinces in the South. In columns 3 and 4, the sample includes eight "Destination" provinces in the South-those with highest shares of Northernborn immigrants. The sample is restricted to individuals aged 20-64 from 1954-1989 birth cohorts. All columns control for age in cubic polynomial, education level, ethnicity, province fixed effects, gender-by-year fixed effects, gender-by-birth-origin fixed effects, and interactions of provinces' initial characteristics with year dummies (as in column 3 of Table 2). Columns 1 and 2 control for for gender-by-birth-origin-by-destination fixed effects. Birth origin is an indicator for being born in the North. Destination is an indicator for living in a Destination province. Standard errors are clustered at the province level, * p < 0.10, ** p < 0.05, *** p < 0.01.



Figure A1: Export value by industry, 2002-2018 (trillion VND, constant 2018 price)

Notes: Unit in Trillion VND, constant 2018 price. Source: UN Comtrade.



Figure A2: Average years of schooling by gender and region

Source: Vietnam Population and Housing Census 2009.



Figure A3: Number of kindergarten classes and teachers by region

Notes: The dash lines use figures apportioned to each region using country-level data, applying region shares in 1985 for 1986-1987, 1988 for 1989-1992, and 1996 for 1993-1995. Source: General Statistics Office of Vietnam.



Figure A4: Industry employment composition by gender over time

Notes: The sample is restricted to working individuals aged 20-64. Source: VHLSS 2002-2018.

Figure A5: Opinions of Southern Vietnamese compared to Northern Vietnamese by gender



Notes: Coefficients come from separate regressions of each outcome (opinion) on the South indicator, controlling for year fixed-effects and separately for each gender. Answers are recoded from "Strongly Agree", "Agree" and "Neither" into 1 (Agree); "Disagree" and "Strongly Disagree" into 0 (Disagree). Source: World Values Survey 2001, 2006 and 2019.

Figure A6: Employment share by region in 1999

(a) 1-digit industry employment structure by region in 1999



(b) 2-digit manufacturing employment share by region in 1999



Notes: Predicted export exposure is constructed according to equation (1). Export values (constant 2018 VND) from United Nations Comtrade Database (UN Comtrade) are adjusted for the Consumer Price Index obtained from the General Statistics Office (GSO) of Vietnam. In Panel B, the solid purple line represents the 17th parallel north, the Demilitarized Zone dividing North and South Vietnam.

Figure A7: Household size



Source: VHLSS 2002-2018.





Source: Vietnam Population and Housing Censuses 1999, 2009 and 2019.



Figure A9: Labor force participation rate by gender and birth origin, South Vietnam subsample

Notes: This figure plots the share of southern individuals aged 20-64 who reported to have worked in the last 12 months, separately by birth origin and gender. Southerners born in the North are people whose birth province belongs to former North Vietnam territory. Source: VHLSS 2014 – 2018.

Appendix B. Plausibility of Share Exogeneity

In Section 3, we argue that our identifying assumption, i.e., the exogeneity of the shift-share instrument in our main equation, stems from "exogenous shares" (Goldsmith-Pinkham et al., 2020). We support this assumption using the diagnostic tests suggested in Goldsmith-Pinkham et al. (2020) and Borusyak et al. (2025).

We first aggregate the individual-level outcomes to the province level and estimate a province-level analogue of equation (2):

$$\bar{y}_{pt} = b_0 + b_1 Exposure_{pt} + \phi_p + \psi_t + \bar{X}_{pt}\lambda + Z_p\gamma_t + u_{pt}, \quad (2a)$$

which can also be expressed in first differences:

$$\Delta \bar{y}_{pt} = b_1 \Delta Exposure_{pt} + \Delta \psi_t + \Delta \bar{X}_{pt} \lambda + Z_p \Delta \gamma_t + \Delta u_{pt}. \quad (2b)$$

Following Goldsmith-Pinkham et al. (2020), we then determine the five industries with the highest Rotemberg weights, i.e., industries most important for determining variation in $Exposure_{pt}$, to examine the variation in the data that the estimator is using. As reported in **Table B1**, these five instruments account for 79 percent (0.831/1.049) of the positive weight in the estimator.

We visually inspect the dispersion of point estimates across instruments in **Figure B1**, which illustrates the heterogeneity of these estimates against the first-stage F-statistics. We find that the point estimates are fairly consistent across instruments, with minimal dispersion. Moreover, the high-weight industries are closely aligned with the overall point estimate. Focusing on the five key industries, we conduct two additional diagnostic tests to assess the plausibility of the identifying assumption.

Pretends. Our analysis focuses on the period between 2002 and 2018, when Vietnam's export was expanding rapidly. To check potential "pretrends," we obtained employment data from VHLSS 1998. Specifically, we check whether the change in a province's employment rate in

the pre-shock period ($\Delta \bar{y}_{pt}$ between 1998 and 2002) may correlate with the employment shares of the five key industries or the shift-share variable itself ($\Delta Exposure_{pt}$ between 2002 and 2018), using equation (2b). Regressions are weighted by province's 1999 population.

As shown in the first panel of **Table B2**, we do not find significant correlations between employment growth during the pre-period (1998-2002) and the industry shares (columns 1 through 5). Moreover, employment growth during the pre-period is uncorrelated with future growth in export exposure (column 6). In contrast, when looking at the study period (2002-2018), in the second panel, most correlations between industry shares and employment growth during the study period (2002-2018) are negative. In particular, the estimate in column 6 is the province-level first-difference analogue of our baseline results (column 2 of Table 2). The point estimate is the same (-0.013), however, it is less precisely estimated due to using only the initial and final years (omitting the intermittent years used in the main analysis).

Correlates of Initial Industry Composition. Next, we examine the correlates of the industry employment shares in the initial period. In **Table B3**, we examine the correlation between the employment shares of the five industries and key provincial characteristics in 1999 which we believe may be correlated with the innovation term in determining employment growth (Δu_{pt}) . In the last column, we also report the relationship between these covariates and $\Delta Exposure_{pt}$ (between 2002 and 2018). These characteristics explain a fair amount of the cross-sectional variation in the shares and the shift-share variable itself.

In all regressions in the paper, we control for these observables. Moreover, in Table 2, the inclusion of these covariates shifts the estimate away from zero (from -0.007 in column 1 to -0.013 in column 2). This suggests that unobserved confounders, if having similar effects as these observables, would make the estimate even more negative. We formally assess the extent of bias in our estimates by conducting the Oster test (Altonji et al, 2005; Oster, 2019). The

results are provided in **Table B4**. Panel A shows results for the individual-level regression while Panel B shows results for the province-level regression. Columns 1 and 2 indicate movement in coefficients as we include controls for province initial characteristics (interacted with year dummies). If we assume that the unobservable confounders are as important as the observable (i.e., set $\delta = 1$ in the framework of Oster (2019)), then the effect of export exposure becomes even more negative, see columns 3 and 4 (under different assumptions of R-squared movement).

References

- Altonji, Joseph G., Todd E. Elder, and Christopher R. Taber. 2005. "Selection on observed and unobserved variables: Assessing the effectiveness of Catholic schools." *Journal of Political Economy* 113 (1): 151-184.
- Borusyak, Kirill, Peter Hull, and Xavier Jaravel. 2022. "Quasi-Experimental Shift-Share Research Designs." *The Review of Economic Studies* 89 (1): 181–213
- Borusyak, Kirill, Peter Hull, and Xavier Jaravel. 2025. "A Practical Guide to Shift-Share Instruments." *Journal of Economic Perspectives* 39 (1): 181–204.
- Goldsmith-Pinkham, Paul, Isaac Sorkin, and Henry Swift. 2020. "Bartik Instruments: What, When, Why, and How." *American Economic Review* 110 (8): 2586–2624.
- Oster, Emily. 2019. "Unobservable Selection and Coefficient Stability: Theory and Evidence." Journal of Business and Economic Statistics 37 (2): 187–204.

Panel A: Negative and positive weights							
	Sum	Mean	Share				
Negative	-0.049	-0.003	0.045				
Positive	1.049	0.022	0.955				
Panel B: Correlations of Industry Aggregates							
	α_k	g_k	β_k	F_k	$\operatorname{Var}(z_k)$		
$lpha_k$	1						
g_k	0.675	1					
$eta_{m k}$	0.009	-0.004	1				
F_{k}	0.692	0.378	0.007	1			
$\operatorname{Var}(z_k)$	-0.056	-0.076	0.002	-0.071	1		
Panel C: Top 5 Rotemberg weight industries							
	\hat{lpha}_k	g_k	$\hat{eta}_{m k}$	$95~{\rm percent}~{\rm CI}$	Ind. Share		
TV, radio transmitters, line telephony	0.366	1433.805	-0.014	(-0.026, 0.006)	0.013		
Crude petroleum and natural gas	0.268	759.045	-0.022	(-0.038, -0.003)	0.015		
General purpose machinery	0.097	331.500	-0.021	(-0.050, 0.003)	0.033		
Office, accounting, computing machinery	0.056	1978.641	-0.019	(-0.050, 0.050)	0.003		

Panel A: Negative and positive weights

TV, radio receivers, recording apparatus

Panel D: Estimates of β_k for positive and negative weights

r	α -weighted		
Negative	0.003	-0.186	-0.129
Positive	-0.017	1.186	0.020

460.477

-0.005

(-0.022, 0.039)

0.012

0.044

This table reports statistics about the Rotemberg weights. In all cases, we report statistics about the aggregated weights with normalized growth rates, where we aggregate a given industry across years and normalize growth rates to the per-period average. Panel A reports the share and sum of negative weights. Panel B reports correlations between the weights $(\hat{\alpha}_k)$, the national component of growth (g_k) , the just-identified coefficient estimates $(\hat{\beta}_k)$, the first-stage F-statistic of the industry share (F_k) , and the variation in the industry shares across locations $(Var(z_k))$. Panel C reports variation in the weights across years. Panel C reports the top five industries according to the Rotemberg weights. The g_k is the national industry growth rate, $\hat{\beta}_k$ is the coefficient from the just-identified regression, the 95 percent confidence interval is the weak instrument robust confidence interval using the method from Chernozhukhov and Hansen (2008) over a range from -10 to 10, and Ind. Share is the industry share (multiplied by 100 for legibility). Panel D reports statistics about how the values of $\hat{\beta}_k$ vary with the positive and negative Rotemberg weights.

	(1)	(2)	(3)	(4)	(5)	(6)
	TV, radio transmitters, line telephony	Crude petroleum and natural gas	General purpose machinery	Office, accounting, computing machinery	TV, radio receivers, recording apparatus	Δ Exposure
Pre-exposure						
$\Delta Work^{1998-2002}$	0.101	0.035	0.100	0.167	0.512	-0.002
	(0.231)	(0.055)	(0.079)	(0.730)	(0.388)	(0.012)
Current period						
$\Delta Work^{2002-2018}$	-0.204	0.035	-0.048	-1.142^{**}	-0.334	-0.013*
	(0.147)	(0.035)	(0.052)	(0.435)	(0.239)	(0.007)

Table B2: Relationship between industry shares and trends in work

Notes: Each coefficient comes from a separate regression of change in average provinces' working rate on a 1999 industry share (columns 1 to 5) or on change in Exposure during 2002-2018 (column 6). All specifications control for provinces' initial characteristics and end-of-period demographic background (average age, share female, share population by education level, share major ethnicity at the province level). Results are weighted by provinces' initial population in 1999. For legibility, coefficients and standard errors of the first five columns are divided by 100. Coefficients and standard errors of the last column is not scaled. Standard errors in parentheses, * p < 0.10, ** p < 0.05, *** p < 0.01.

Table B3: Relationship between industry shares and provinces' initial characteristics

	TV, radio transmitters, line telephony	Crude petroleum and natural gas	General purpose machinery	Office, accounting, computing machinery	TV, radio receivers, recording apparatus	Δ Exposure
Share Mfg. Emp.	0.000 (0.001)	-0.005 (0.005)	-0.001 (0.002)	$0.000 \\ (0.000)$	$0.001 \\ (0.001)$	0.014 (0.021)
Mfg output	0.004 (0.006)	$0.055 \\ (0.046)$	$0.028 \\ (0.018)$	0.004 (0.002)	$0.006 \\ (0.004)$	0.117 (0.152)
Service output	0.029 (0.021)	-0.023 (0.030)	$0.002 \\ (0.022)$	-0.003 (0.006)	$0.010 \\ (0.011)$	$0.194 \\ (0.238)$
Female LFP	0.078 (0.053)	-0.486 (0.418)	$\begin{array}{c} 0.125 \\ (0.153) \end{array}$	$0.030 \\ (0.013)$	$0.062 \\ (0.032)$	3.840 (1.495)
Kindergarten	$0.002 \\ (0.001)$	0.003 (0.002)	-0.001 (0.001)	-0.000 (0.000)	$0.001 \\ (0.001)$	0.003 (0.014)
R^2 Observations	$\begin{array}{c} 0.432 \\ 60 \end{array}$	$\begin{array}{c} 0.261 \\ 60 \end{array}$	$\begin{array}{c} 0.136\\ 60 \end{array}$	$\begin{array}{c} 0.418\\ 60\end{array}$	$\begin{array}{c} 0.613 \\ 60 \end{array}$	0.264 60

Notes: Each of the first five columns reports results of a single regression of a 1999 industry share on 1999 characteristics. The last column reports results when regressing change in Exposure during 2002-2018 on 1999 characteristics. Results are weighted by provinces' initial population in 1999. For legibility, coefficients and standard errors of the first five columns are multiplied by 100. Coefficients and standard errors of the last column is not scaled. Standard errors in parentheses.

$\beta_{baseline}$	$\beta_{controlled}$	$\beta_{bias-adjusted}$	$\beta_{bias-adjusted}$
	(Std. error) \tilde{z}	-max - ()	if $R_{max}^2 = min(1, 1.3\tilde{R})$
L J	$[R^2 = \tilde{R}]$	$[R_{max}^2]$	$[R_{max}^2]$
(1)	(2)	(3)	(4)
Panel A. Ir	ndividual lev	el	
-0.007	-0.013	-0.069	-3.407
(0.005)	(0.004)		
[0.128]	[0.129]	[0.130]	[0.168]
Panel B. P	rovince level		
-0.012	-0.013	-0.015	-0.150
(0.005)	(0.005)		
[0.887]	[0.901]	[0.916]	[1.000]

Table B4: Oster test for coefficient stability

Notes: The dependent variable is "Work" for the individual-level regression in Panel A, and average provinces' working rate for the province-level regression in Panel B. Column 1 shows the baseline effects. Baseline controls include demographic background (age in cubic polynomial, gender, education level, ethnicity in Panel A, and average age, share female, share population by education level, share major ethnicity at the province level in Panel B), province fixed effects, year fixed effects, and interaction of provinces' initial share of manufacturing employment with year dummies. Column 2 shows the estimated effects in our full model when additionally controls for interactions of provinces' initial characteristics with year dummies. The estimates in column 1 and column 2 of Panel A correspond to column 1 and column 2 of Table 2 in the main text, respectively. Columns 3 and 4 show the bias-adjusted effect under the assumption on R^2_{max} specified in the column header and $\tilde{\delta} = 1$.

Figure B1: Heterogeneity of β_k



○ Positive Weights ◇ Negative Weights

Notes: This figure plots the relationship between each instrument's β_k , first-stage F-statistics and the Rotemberg weights. Each point is a separate instrument's estimate (industry share). The figure plots the estimated β_k for each instrument on the y-axis and the estimated first-stage F-statistic on the x-axis. The size of the points are scaled by the magnitude of the Rotemberg weights, with the circles denoting positive weights and the diamonds denoting negative weights. The horizontal dashed line is plotted at the value of the overall $\hat{\beta}$ estimated at the province level. The figure excludes instruments with first-stage F-statistics below 5.